THE EFFECTS OF DIVERSION TERRACES, GRASSED WATERWAYS, AND TILE DRAINAGE ON SELECTED SOIL PROPERTIES AND GREENHOUSE GAS EMISSIONS

by

Bryan Andrew Driscoll

B.Sc., The University of British Columbia, 2016

A THESIS SUBMITTED IN PARTIAL FULFILLMENT OF

THE REQUIREMENTS FOR THE DEGREE OF

MASTER OF SCIENCE

in

THE FACULTY OF GRADUATE AND POSTDOCTORAL STUDIES

(Soil Science)

THE UNIVERSITY OF BRITISH COLUMBIA

(Vancouver)

August 2022

© Bryan Andrew Driscoll, 2022

The following individuals certify that they have read, and recommend to the Faculty of Graduate and Postdoctoral Studies for acceptance, a thesis entitled:

The Effects of Diversion Terraces, Grassed Waterways, and Tile Drainage on Selected Soil Properties and Greenhouse Gas Emissions

submitted by	Bryan Driscoll	in partial fulfillment of the requirements for
the degree of	Master of Science	
in	Soil Science	

Examining Committee:

Dr. Maja Krzic, Associate Professor, Faculty of Land and Food Systems / Faculty of Forestry, UBC Supervisor

Dr. Louis-Pierre Comeau, Adjunct Professor, Faculty of Land and Food Systems, UBC Supervisory Committee Member

Dr. Bianca Eskelson, Associate Professor, Faculty of Forestry, UBC Supervisory Committee Member

Dr. Cindy Prescott, Professor, Faculty of Forestry, UBC Additional Examiner

Abstract

New Brunswick is one of the largest potato producing regions in Canada, which has experienced soil erosion and a decline in soil health due to intensive tillage from potato production. To address these concerns and the uncertainties of temperature and precipitation pattern changes from climate change, a new system has been proposed for soil and water conservation in Atlantic Canada. The landscape integrated soil and water conservation (LISWC) system combines diversion terraces, grassed waterways, tile drainage, water retention structures, supplemental irrigation, conservation tillage, and soil-landscape restoration. The objective was to evaluate the effects of diversion terraces, grassed waterways, and tile drainage relative to contour tillage on selected soil properties and greenhouse gas emissions over the initial three seasons of management implementation. The study site consisted of three land management practices (LMPs): contour tillage (CT), diversion terraces and grassed waterways (DTGW), and diversion terraces, grassed waterways, and tile drainage (DTGW+TD). Aggregate stability decreased in all LMPs, and LMP had a significant effect on labile carbon (POXC). CT led to a decrease of 11.2% POXC over the study period, while DTGW increased 19.8% and DTGW+TD increased 50.6%. Mineral associated organic matter (MAOM) carbon was affected by LMP: CT carbon concentrations decreased 3.75%, while DTGW+TD rose 9.7% and DTGW rose 15.8%.

I also conducted a short-term incubation experiment in February of 2021 to quantify greenhouse gas fluxes in response to volumetric water content (20%, 27%, 35%, and 40%), nitrogen fertilizer application rates (0 and 170 kg N ha⁻¹), and temperature (10°C and 25°C).

Overall, both DTGW and DTGW+TD offer potential for improvement of soil health, relative to CT, in the second and third years of implementation, with increases in the labile C fraction POXC, a fraction significantly related to SOC, and increases in MAOM. This is the first study in the area measuring GHG emissions from erosion control structures. This study demonstrated that DTGW+TD significantly reduced the loss of CO_2 relative to DTGW, while also not emitting significantly more than CT, in the three years after implementation. Results were similar with respect to loss of nitrogen as N₂O.

Lay Summary

Potatoes are an economically important crop in New Brunswick, but they require intensive management. The objective of my study was to assess how soil properties and greenhouse gas emissions are affected by combinations of best management practices – including diversion terraces, grassed waterways, and tile drainage – to form two new land management practices (LMPs) for potato production. No LMP was able to protect against a decline in aggregate stability during the second and third years after establishment, but active carbon increased in both new LMPs. I also found that carbon in the mineral fraction was greater in the new LMPs relative to contour tillage. This represents a more stable form of carbon and could lead to carbon sequestration. More CO₂ and N₂O was emitted from the LMP that excluded tile drainage. This research supports development of new management practices that could improve soil health and resiliency to climate change in the potato industry in this region.

Preface

This thesis represents unpublished work, which I conducted with assistance from undergraduate students, research technicians, and advisors. I was the lead investigator in the studies included in chapters 2, 3, and 4. The work hereafter is the result of a collaboration between me and the Fredericton Research and Development Centre (FRDC). Initial design of the experiment site was done by Dr. Sheng Li of the FRDC, and design of the current experiment was done by Dr. Louis-Pierre Comeau.

Data in chapters 2, 3, and 4 were collected by me and the research team at the FRDC, including Kayli McGarrigle, Kyle MacKinley, Yulia Kupriyanovich, and Dr. Louis-Pierre Comeau. Additional assistance with lab work was provided by Allysia Murphy from the University of New Brunswick and from Camilla Menhardt and Paula Porto from the University of British Columbia.

Dr. Louis-Pierre Comeau was the supervisory author on this project and was closely involved on all aspects of the study and thesis. The study was completed in collaboration with Dr. Maja Kržić and Dr. Bianca Eskelson, who aided in the design of the experiment in Chapter 4 and interpretation of all chapter results. All three were instrumental in providing thesis edits.

Table of Contents

Abstract ii
Lay Summary
Prefacev
Table of Contents
List of Tablesxi
List of Figures xii
List of Abbreviations and Symbolsxvii
Acknowledgementsxix
Dedicationxx
Chapter 1: General Introduction
1.1 Background
1.2 Soil Quality Concerns in New Brunswick
1.2.1 Soil organic carbon
1.2.2 Erosion
1.2.2.1 Soil Water Erosion
1.2.2.2 Tillage Erosion
1.3 Diversion Terraces and Grassed Waterways
1.4 Tile Drainage
1.5 Contour Tillage
1.6 Landscape Integrated Soil and Water Conservation (LISWC)
1.7 Research Objectives and Hypotheses
vi

Tillage, Diversion Terrace, Grassed Waterway, and Tile Drainage Implementation	on12
2.1 Introduction	12
2.2 Materials and Methods	14
2.2.1 Site Description	14
2.2.2 Field Seasons	16
2.2.2.1 2020 Field Season	16
2.2.2.2 2021 Field Season	17
2.2.3 Sampling and Analysis	18
2.2.3.1 Soil Organic Carbon	19
2.2.3.2 Soil Aggregate Stability	19
2.2.3.3 Permanganate Oxidizable Carbon	20
2.2.3.4 Particulate Organic Carbon	20
2.2.4 Statistical Analysis	21
2.3 Results	21
2.3.1 Soil Organic Carbon	21
2.3.2 Soil Aggregate Stability	23
2.3.3 Permanganate Oxidizable Carbon	25
2.3.4 Particulate Organic Carbon	27
2.4 Discussion	31
2.4.1 Soil Organic Carbon	31
2.4.2 Soil Aggregate Stability	31
2.4.3 Permanganate Oxidizable Carbon	32
	viii

Chapter 2: Short-term Response of Soil Aggregate Stability and Labile Carbon to Contour

2.4.4 Particulate Organic Carbon	34
2.5 Conclusions	36
Chapter 3: Greenhouse Gas Emissions and Nitrate Supply Rate Response to Initial	Three
Years of Diversion Terrace, Grassed Waterway, and Tile Drainage Implementation	ı38
3.1 Introduction	
3.2 Materials and Methods	40
3.2.1 Site Description	40
3.2.2 Sampling and Analysis	42
3.2.2.1 CO ₂ Readings	42
3.2.2.2 N ₂ O and CH ₄ Readings	43
3.2.2.3 Nitrate Supply Rate	45
3.2.3 Statistical Analysis	45
3.3 Results	46
3.3.1 Carbon Dioxide	46
3.3.2 Nitrous Oxide	51
3.3.3 Methane	52
3.3.4 Nitrate Supply Rate	57
3.4 Discussion	58
3.4.1 Carbon Dioxide	58
3.4.2 Nitrous Oxide	59
3.4.3 Methane	61
3.4.4 Nitrogen Supply Rate	61
3.5 Conclusions	62
	ix

- · · I	
Emissi	ons in a Short-Term Incubation Experiment64
4.1	Introduction64
4.2	Materials and Methods65
4.2	2.1 Experimental Design
4.2	2.2 Sampling and analysis67
4.2	2.3 Statistical Analysis
4.3	Results
4.4	Discussion72
4.5	Conclusions75
Chapte	er 5: General Conclusions and Recommendations for Future Research77
5.1	Soil Aggregate Stability and Labile Carbon Response to Second and Third Years of
Cont	our Tillage, Diversion Terrace, Grassed Waterway, and Tile Drainage Implementation .77
5.2	Greenhouse Gas Emissions and Nitrate Supply Rate Response to Initial Three Years of
Dive	rsion Terrace, Grassed Waterway, and Tile Drainage Implementation80
5.3	Combined Effects of Nitrogen, Temperature, and Soil Water Content on GHG
Emis	ssions in a Short-Term Incubation81
5.4	Recommendations for Future Research
Bibliog	graphy
Appen	dices103
Appe	endix A Photos of the LISWC system at the FRDC103
A.	1 CT (left), DTGW (middle), DTGW+TD (right) with oats in the field in 2020 103

Chapter 4: Combined Effects of Nitrogen, Temperature, and Soil Water Content on GHG

A.2	View from the side of one of the grassed waterways with the soil berm on the	
downslope edge (left)104		
A.3	View of DTGW+TD plots (foreground) from the side1	05
A.4	View of CT from the bottom looking upward after a heavy rainfall in which visible	e
signs c	of water erosion appear1	06
Appendix	x B List of soil properties for each LMP and slope position. Bulk density was	
determin	ed August 19th and 26th, 2020. All others determined July 2nd, 20201	07
Appendix	x C Timeline of events in 20201	08
Appendix	x D Timeline of events in 20211	09
Appendix	x E Rill erosion after Hurricane Elsa1	11
Appendix	x F Soil Aggregate Stability Analysis Procedure1	12
Appendix	x G Soil Aggregate Stability by LMP in 2020 and 20211	13
Appendix	x H Permanganate Oxidizable Carbon Analysis Procedure1	14
Appendix	x I Density Fractionation Analysis Procedure1	16
Appendiz	x J Diagram of modified protocol for density fractionation1	18
Appendix K Collar set up for collecting GHG samples in 2019, 2020, and 2021119		
Appendiz	x L COVID-19 notes from my workbook1	20

List of Tables

Table 4-1 Pairwise comparisons of volumetric water content (VWC) groups, averaged over N	1
fertilization level, using Tukey adjustment.	70

List of Figures

Figure 2-6 Permanganate oxidizable carbon (POXC) values in CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage). Soil measurements were taken on July 2nd and September 24th, 2020 and May 27th and October 7th, 2021. Boxplots represent one slope position (upper, middle, Figure 2-7 Percentage change (%) in permanganate oxidizable carbon (POXC) between postplanting 2020 and post-harvest 2021 in CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage). Figure 2-8 Organic C concentrations per kg of soil between post-planting 2020 and post-harvest 2021 in CT (contour tillage), DTGW (diversion terraces and grassed waterways), and Figure 2-9 Top: Change in organic C concentration per kg of soil between post-planting 2020 and post-harvest 2021 in CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage). Bottom: Figure 3-1 Schematic of the study site layout and the 3 land management practices (LMP) – CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion Figure 3-2 A) CO₂ fluxes from May 28th to September 22nd, 2020 in the 3 land management practices: CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage). B) Mean daily

Figure 3-3 A) CO₂ fluxes from April 29th to September 28th, 2021 in the 3 land management practices: CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage). B) Mean daily Figure 3-4 Mean daily emissions (kg CO₂-C ha⁻¹) for each LMP – CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage) – in each of the 2020 and 2021 seasons. Error bars represent the standard error of the mean (2020, n = 348; 2021, n = 360). Different letters indicate a statistically Figure 3-5 Total CO₂ emissions (kg C ha⁻¹) in 2020 and 2021 in each LMP – CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage). Error bars represent the standard error of the mean (n = n)12). Different letters indicate a statistically significant difference between LMPs ($\alpha = 0.05$).50 Figure 3-6 Soil temperature data in 2020 and 2021 for each LMP – CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed Figure 3-7 Soil volumetric water content data in 2020 and 2021 for each LMP - CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, Figure 3-8 Mean daily emissions of $N_2O(A)$ and $CH_4(B)$ in 2019, 2020, and 2021, with soil temperature (C) and soil moisture (D), and coloured by LMP - CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed

Figure 3-9 Effect of land management practice (LMP) – CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage) – on emission of $N_2O(A)$ and $CH_4(B)$ across 2019, 2020, and 2021 growing Figure 3-10 Emission of N_2O by slope position in each land management practice – CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, Figure 3-11 Total N₂O emission for the 2019, 2020, and 2021 seasons expressed in CO₂ equivalents (CO₂e) (kg C ha⁻¹) for each LMP – CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage). Error bars represent the standard error of the mean (n = 12). Different letters indicate a statistically significant difference between LMPs ($\alpha = 0.05$), but comparisons can only be made Figure 3-12 Total CH₄ emission for the 2019, 2020, and 2021 seasons expressed in CO₂ equivalents (CO₂e) (kg C ha⁻¹) for each LMP – CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage). Error bars represent the standard error of the mean (n = 12). Different letters indicate a statistically significant difference between LMPs ($\alpha = 0.05$), but comparisons can only be made Figure 3-13 Effect of land management practice – CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage) on nitrate supply rates measured in (top) 2020 and (bottom) 2021. The four sampling periods in 2020 were: (1) June 17 – July 12; (2) July 2 – July 28; (3) July 28 – August

xvi

18; and (4) August 19 – September 8. The three sampling periods in 2021 were: (1) July 26 –
August 16; (2) August 16 – September 7; and (3) September 7 – September 22. Error bars
represent the standard error of the mean $(n = 12)$
Figure 4-1 Mean CO ₂ flux calculated at the end of the 6-day incubation experiment. Error bars
represent the standard error of the mean $(n = 9)$
Figure 4-2 Mean N ₂ O flux calculated at the end of the 6-day incubation experiment. Error bars
represent the standard error of the mean $(n = 9)$. For each volumetric water content (VWC)
within a temperature, asterisks denote a significant difference between nitrogen fertilizer rate \times
VWC treatments ($\alpha = 0.05$)
Figure 4-3 Mean CH ₄ flux calculated at the end of the 6-day incubation experiment. Error bars
represent the standard error of the mean $(n = 9)$
Figure 4-4 Total CO ₂ equivalent emissions over the 6-day incubation experiment. Error bars
represent the standard error of the mean $(n = 9)$

List of Abbreviations and Symbols

Ap – An A horizon that has been disturbed by human activity BMP - Best management practice C – Carbon $CH_4 - Methane$ CO₂ – Carbon dioxide CO₂ eq – Carbon dioxide equivalent DOC – Dissolved organic carbon DTGW - Diversion terrace and grassed waterway fPOM – Free light particulate organic matter FRDC - Fredericton Research and Development Centre GHG - Greenhouse gases GWC - Gravimetric water content LMP - Land management practice MAOM - Mineral associated organic matter MWC - Maximum water capacity NaN – Not a number NB – New Brunswick $NH_4^+ - Ammonium$ $NO_3^- - Nitrate$ N₂O - Nitrous oxide oPOM - Occluded particulate organic matter PPM – Parts per million POM - Particulate organic matter POX-C - Permanganate oxidizable carbon SAS – Soil aggregate stability SCE – Soil CO₂ efflux SOC – Soil organic carbon SOM - Soil organic matter SWC – Soil water content TD – Tile drainage VWC – Volumetric water content WFPS – Water-filled pore space

Acknowledgements

I would like to sincerely thank the members of my graduate committee for their support and guidance during my degree. Dr. Maja Kržić was my academic advisor and, along with Dr. Sandra Brown, was responsible for inspiring me during undergraduate studies to pursue soil science further. Dr. Louis Pierre Comeau was my research supervisor in Fredericton. Unfortunately, due to the Covid-19 pandemic we did not get to see each other much in person. But the times we had sampling and potato harvesting were fun. Thanks for giving me the opportunity to work at the FRDC and explore New Brunswick for two years. Finally, thank you to Dr. Bianca Eskelson for your help with stats and making the stats course manageable and fun.

Perhaps the biggest thanks go out to the two field technicians at the FRDC who were immensely helpful and supportive in my summers in Fredericton. Thank you, Kyle MacKinley. Thank you, Yulia Kupriyanovich. Every grad student should have a Kyle and a Yulia!

Thanks go out to Raelani Kesler, who started the same program in 2019 with me. It was a pleasure to travel this road with you. The bike rides, the runs, and the potluck dinners were all a cherry on top. We did it!

Special thanks are owed to my parents, who have supported me throughout my years of education. Once farmers, always farmers. Love you \clubsuit

I would like to acknowledge that my research in Fredericton, New Brunswick took place on the traditional territories of the Wolastoqiyik, Mi'kmaq, and Peskotomuhkati peoples.

Dedication

This thesis is dedicated to my late grandmother, Ina Brownell, who generously offered her financial support when I first decided to go to university. Thanks, Gram!

Chapter 1: General Introduction

1.1 Background

Soil and water are fundamental natural resources for potato (*Solanum tuberosum* L.) production in Atlantic Canada, where Prince Edward Island and New Brunswick alone accounted for about one third of all Canadian land seeded with potato in 2019 (Agriculture and Agri-Food Canada, 2020). Potatoes are the main crop grown in the province of New Brunswick. There were over 21,000 hectares of land in the province seeded with potato in 2020, representing 166 million dollars in farm cash receipts (Statistics Canada, 2021).

Potato production requires intensive management, causing high soil disturbance from numerous tillage operations like disking, harrowing, planting, hilling, and harvesting (Wilson et al., 2018). It also requires numerous spray applications of fungicide and insecticide to battle blight and Colorado potato beetle, respectively, as well as a topkill application two weeks before harvest (Govaere et al., 2019). Furthermore, the amount of crop residue left after harvest is low (Zebarth et al., 2022). This means there is little to protect the soil surface from erosive wind and water forces, as well as little in the way of organic matter to be incorporated back into the soil. Potatoes are also a high nutrient uptake crop, and, because of all these factors, are known as a soil-depleting crop (Government of New Brunswick, n.d. -a).

It is expected that climate change will bring increases in precipitation, weather variability, and occurrence of extreme weather events to Atlantic Canada (Stocker et al., 2013; Hoegh-Guldberg et al., 2019). Potato fields in this region are subject to heightened soil erosion from water runoff and further intensified by tillage (Chow et al., 1990; 2000). This, in turn, leads not only to loss of fertile soil, but groundwater pollution by leached chemicals, pathogens, and

nutrients (Li, 2021). To effectively address these challenges and to conserve soil and water resources in this region, there is a need to develop and refine best management practices (BMPs).

1.2 Soil Quality Concerns in New Brunswick

1.2.1 Soil organic carbon

New Brunswick is entirely situated within the Appalachian Mountain range – a chain of old, eroded mountains that has given rise to river valleys and low, gently rolling hills. Much of the potato production in the province happens to the northwest, in an area known as the 'potato belt', close to the border with Maine. Soils are generally characterized by poor structure and high bulk density, which is even more prominent at lower depths (Chow, 1999). The slopes in this area generally range from 2% to 9%, but in rare cases can be above 15% (Stuart, 2017). This makes it a topographically complex landscape for farming – particularly for intensive crops such as potatoes that require constant field activity.

Soil health and soil loss have been of great concern in the region for decades; soil organic matter (SOM) content in intensive potato production fields in New Brunswick has been declining since the 1960s (Eilers et al., 2010; Rees et al., 2014). Over the past 40 years or more, economic pressures in North America have resulted in increased crop specialization and greater potato production intensity (Stark & Porter, 2005). The soil disturbances that accompany potato production often break down soil aggregates, oxidize organic matter, and increase erosion (Lal et al., 1994; Unger, 1992). The result is a decline in SOM and soil organic carbon (SOC) – often-used indicators of soil health.

Over the years, crop rotations have gotten shorter, and the frequency of potato production has increased; a typical rotation is potato-potato-grain or potato-grain (Rees et al., 1999). These rotations do not provide adequate organic carbon (C) inputs to maintain or improve SOC levels

(Zielke & Christenson, 1986). For example, between 1989 and 1999, Rees et al. (2007) reported an 8% reduction from a long-term potato benchmark site under intensive up-and-down slope cultivation in northwestern New Brunswick – near Grand Falls. In addition, Rees et al. (2008) also reported a 4% reduction in SOC in a paired benchmark site under intensive potato production in a variable grade diversion and grassed waterway system in the same region. In 1984, Wang et al. reported an average of just 17.1 g SOC kg soil⁻¹ in loam-textured plow layers of intensively cropped potato belt soils. Prior to the intensification of production, Langmaid et al. (1976) found 26.3 to 47.8 g C kg soil⁻¹ in the Ap horizons of similar potato belt soils. An SOC concentration of 20.0 to 29.0 g kg⁻¹ has previously been reported as the ideal range for Atlantic Canadian potato production (Hinds, 1989). Organic matter contributes significantly to the soil's ability to store and exchange nutrients. With a decline in SOM, the likelihood of lowered nutrient retention is to be expected (Stark & Porter, 2005).

1.2.2 Erosion

1.2.2.1 Soil Water Erosion

Soil erosion, a process which removes topsoil and deposits it elsewhere, is also a major concern for Atlantic Canada. Because of high levels of precipitation, potato crops are predominantly rain-fed in Atlantic Canada (Li, 2021). The area receives high-intensity rainfall events during the summer months and freeze-thaw cycles during winter months. Earlier research has identified water erosion as a serious problem (Chow et al., 1999; Edwards et al., 1998; Saini & Grant, 1980; Wang et al., 1984). Coote et al. (1981) reported that soil water erosion in northwestern New Brunswick was among the most severe seen in all of Canada. The risk of water runoff and runoff-induced soil erosion are high when row crops are located on hilly landscapes with shallow glacial till soils and in areas with high levels of precipitation. Annual

soil loss from runoff erosion in the mid-1990s has exceeded 24 Mg ha⁻¹ in up-and-down slope cultivated study plots in drainage basins of the Upper Saint John River Valley (Chow et al., 2010). The degree of erosion will be dependent on water infiltration rate, water holding capacity, structural stability, and intensity and duration of precipitation (Government of New Brunswick, n.d. -b). Because New Brunswick is close to large bodies of water such as the Bay of Fundy and the Atlantic Ocean, erosion can lead to permanent loss of soil, whereas in other regions, like the Prairies, soil erosion leads to soil redistribution.

1.2.2.2 Tillage Erosion

In the last two decades, much research has been done on the effects of tillage erosion. Studies have linked tillage operations to soil erosion and demonstrated that they are significant (Govers et al., 1994; Lindstrom et al., 1990; Lobb et al., 1995). In some studies, tillage erosion has been the dominant form of erosion, exceeding effects of soil redistribution by water (Govers et al., 1996; Quine et al., 1997; Van Oost et al., 2005). In a study at a soil quality monitoring benchmark site, set up by Agriculture and Agri-Food Canada in the New Brunswick potato belt, Tiessen et al. (2009) reported that the "directional movement of soil by tillage operations had been the dominant soil redistribution process" from 1990 to 2005. They estimated the annual average soil loss to be 13.6 Mg ha⁻¹ yr⁻¹, with some field areas exceeding 25 Mg ha⁻¹ yr⁻¹. It is estimated that over half of the cropland in New Brunswick and 75% of the cropland across Atlantic Canada is at risk of unsustainable levels of soil loss (>6 Mg ha⁻¹ yr⁻¹) by tillage erosion (Government of Canada, n.d. -c).

The ability to retain soil is key to also maintaining the productivity and fertility of soil, as erosion can lead to loss of nutrients and organic matter as well as declines in crop yield (Ochuodho et al., 2013). Bakker et al. (2004; 2005) found that on average, a 4% loss in crop

productivity occurred for each 10 cm of soil lost in mechanized agricultural systems. Design and type of plowing instrument, depth and speed of tillage, number of tillage passes, and tillage direction and pattern can all influence the degree and severity of erosion. Researchers in New Brunswick's potato belt have found that a combination of diversion terraces and grassed waterways can help address the control of erosion in the Atlantic region (Chow, 1999).

1.3 Diversion Terraces and Grassed Waterways

Diversion terraces are an engineering means to reducing runoff and soil loss (Agriculture and Agri-Food Canada, 2012; Li, 2021). They are meant to break up long, sloping fields into shorter field segments, thus reducing the kinetic energy of runoff water and, in turn, its erosive power. In the context of Atlantic Canada, the term 'terrace' typically involves a ~1 m high, grassed berm that runs across a field, following the natural contours of that field. A series of berms will be spaced at appropriate intervals (minimum 36 m) and run parallel to each other despite any natural hills or depressions within the field. Diversion terraces (DT) and grassed waterways (GW) are generally used in conjunction with each other. Grassed waterways are usually constructed in natural depressions where water would normally flow or collect. Grassed waterways are built on the upslope side of the berm. The waterways – with a reduced grade – would intentionally be seeded with grasses to slow down surface water runoff, allow increased water infiltration, filter out eroded soil particles, and reduce erosion (Huffman et al., 2013; Government of New Brunswick, n.d. -c). Water is channeled safely out of the field, usually via another grassed waterway running downslope to the base of the field; when the speed of surface runoff is reduced, more water can infiltrate the soil, thereby increasing surface water retention via storage of precipitation and snow melt water (Agriculture and Agri-Food Canada, 2012). It can also increase water availability due to the storage of water in the frontal channel of the

terrace, conserving water for plant growth (Baryla & Pierzgalski, 2008). Research has shown that DTs have been able to reduce runoff and soil loss by 87% and 95%, respectively (Chow et al., 2010).

These BMPs have become more prevalent in recent decades. In the 1990s there was an incentive under the Land Management and Conservation Program to help fund 66% of the costs of implementing diversion terraces and grassed waterways (Rolfe, 1993). The Black Brook Watershed, northeast of the city of Grand Falls, New Brunswick encompasses 1,450 hectares of land. Roughly two thirds of that land is agricultural, and more than 50% of the agricultural land (~460 ha) is now farmed using diversion terraces and grassed waterways (Yang et al., 2009). Statistics for the rest of the province are not as easy to find, but, anecdotally, terraces can be seen all along the Trans-Canada Highway driving between Fredericton and Grand Falls.

Costs to introduce a conservation practice are of considerable importance for farmers. The typical DT takes up 10% of the field being protected, while the typical grassed waterway takes up 4% of the field it protects (Agriculture and Agri-Food Canada, 2012). While terraces will guard against excessive soil loss, it will mean a considerable amount of land being taken out of production. Expected increases in yield may not offset the costs of construction and maintenance. In 2008, annual maintenance costs were pegged at \$84 and \$70 CAD per hectare for DT and GW, respectively. Short term analysis of yield determined no consistent improvements on yield that would offset maintenance costs.

While the Chow et al. (2010) study proved terracing can be extremely effective in preventing soil loss, terraces are not without shortcomings. In the same study, for instance, the researchers found SOC decreased a steady 0.36% each year over a 10-year period – though still

better than the 0.7% drop in the same period from paired up-and-down slope cultivated sites where potato rows were oriented perpendicular to the slope.

Because of the increased water infiltration, there are concerns of fields being too wet in years with high precipitation (Chow et al., 2010). This raises issues with trafficability of the field, which could delay planting and tillage operations or potentially affect yield (Madramootoo et al., 2007). There is also the concern that terraces could induce the movement of chemicals, nutrients, and sediments into surface or groundwater systems, as a greater proportion of typical overland flow would be directed downwards into the soil profile (Li, 2021; Yang et al., 2009).

1.4 Tile Drainage

Tile drains (TD) are an engineering means for removing the excess water after high precipitation events and for managing high water tables (Huffman et al., 2013). Wet field conditions can lead to reduced field trafficability, delayed tillage operations, and, occasionally, yield losses (Madramootoo et al., 2007). In New Brunswick, soils tend to be shallow and are underlain frequently with compact subsoil or low-permeability parent materials (Burton et al., 2012). This can lead to slow subsurface water drainage and waterlogged soil. Though tile drainage is usually installed in low-lying areas with high water tables, they are increasingly being installed in areas that have poor internal drainage (Li, 2021). Adoption in New Brunswick has been rising since the 1970s (Milburn & Gartley, 1988; Ritter et al., 1995). There are many benefits that arise from TD, including increased soil temperature, water infiltration, field trafficability, and crop yield. It can also reduce soil erosion, waterlogging, soil compaction, and phosphorus and potassium losses (Fraser & Fleming, 2001).

The main drawback from TD arises from its role in transporting substances into the soil subsurface. Tile drainage accelerates and provides a shortcut for water flowing from fields to

surface water systems (Li, 2021). The result is a higher risk of contaminants ending up in surface water including, but not limited to, nitrate (NO_3^-), dissolved organic carbon, dissolved nitrous oxide (N_2O), pesticides, and soil sediment (Burton et al., 2012; Milburn et al., 1990).

1.5 Contour Tillage

Contour tillage is one example of a conservation tillage practice that is intended to reduce negative impacts of tillage on soil (Li, 2021). It is an easily implemented practice that orients the ridges and furrows associated with potato production along the natural contours of the field, generally perpendicular to the slope. Conventional tillage for New Brunswick potatoes involves orienting rows up-and-down slope. While this is generally easier and faster to manage with tractors, it increases surface water flow in channels and enhances water erosion. Contour tillage creates barriers that reduce this surface water runoff and water erosion. If a field has drainage problems, this can, however, be an issue.

1.6 Landscape Integrated Soil and Water Conservation (LISWC)

The Landscape Integrated Soil and Water Conservation (LISWC) system was designed in 2017 by Dr. Sheng Li of the FRDC to help conserve soil and water resources on sloping landscapes in Atlantic Canada. The LISWC system is a holistic land management practice (LMP) that includes several types of best management practices (BMPs) (Li, 2021). Individual BMPs included in the LISWC system are DT, GW, TD, water retention structures, supplemental irrigation, conservation tillage practices, and soil-landscape restoration. Each individual BMP has a particular targeted effect and is usually studied on its own. While each BMP may be successful in its own regard, none alone is sufficient to address all soil and water conservation issues facing potato farming. The LISWC strategy is to offset the shortcomings of one BMP by integrating several of them into a single LMP, with the idea that drawbacks from one BMP are

complemented by benefits from another BMP. For example, water runoff and erosion would be reduced by the diversion terraces and grassed waterways, in addition to the implementation of contour tillage. Saturated soil conditions would be addressed by tile drainage. Tile-drained water and runoff would be stored in the water retention structure for potential use later via supplemental irrigation in drought conditions. The hope is that the LISWC system boosts soil health that has been declining for decades in New Brunswick (Eilers et al., 2010; Rees et al., 2014) and increases potato yields that have been stagnant for years (Statistics Canada, 2021). The economic and environmental outcomes of the LISWC system are, however, uncertain since this is a relatively new concept. The aim of my study was to assess how certain soil health indicators (i.e., soil aggregate stability and labile soil carbon fractions) respond to the implementation of the LISWC system. In addition, an aim was to evaluate how higher amounts of rainfall being directed through the soil in diversion terraces affect greenhouse gas (GHG) emissions.

1.7 Research Objectives and Hypotheses

Study objective 1 was to evaluate effects of two LMPs (each comprised of several individual BMPs) relative to contour tillage (CT) on soil aggregate stability across three slope positions in the second and third years after LMP implementation. The two LMPs included the following combinations of BMPs: (i) DT, GW, and CT [from now on labeled DTGW] and (ii) DT, GW, CT, and TD [from now on labeled DTGW+TD].

Hypothesis 1: Soil aggregate stability will decline in DTGW and DTGW+TD,
relative to CT due to the increased soil moisture. The effects will be more severe in
DTGW, since there is no tile drainage to remove excess water after a heavy rainfall. *Objective 1 is addressed in Chapter 2 of this thesis.*

Study objective 2 was to evaluate effects of two LMPs (each comprised of several individual BMPs), relative to CT on total carbon and labile soil organic matter fractions [permanganate-oxidizable C (POXC) and density fractionation of physically un-complexed light fraction organic matter] across three slope positions in the second and third years after LMP implementation. The two LMPs included the following combinations of BMPs: (i) DT, GW, and CT [from now on labeled DTGW] and (ii) DT, GW, CT, and TD [from now on labeled DTGW+TD].

Hypothesis 2: Total carbon will not change in DTGW and DTGW+TD relative to CT over the course of the study. Total carbon changes slowly and therefore will not change over two seasons.

Hypothesis 3: POXC fractions will be lower in DTGW and DTGW+TD relative to CT where soil water content will be higher, thus leading to enhanced decomposition rates.

Hypothesis 4: Free light particulate organic matter (fPOM) will be lower in DTGW and DTGW+TD, relative to CT, as higher amounts of soil water in CT will lead to more rapid decomposition rates.

Hypothesis 5: Mineral-associated organic matter (MAOM) will be adversely affected by DTGW and DTGW+TD, relative to CT. DTGW and DTGW+TD divert more water down into the soil profile. Since MAOM is made up of single molecules and microscopic fragments of OM, more of this OM will be leached out of the soil profile.

Objective 2 is addressed in Chapter 2 of this thesis.

Study objective 3 was to evaluate the effects of DTGW and DTGW+TD, relative to CT, on GHG emissions and nitrate supply rates across three slope positions from the soil during the summers of the initial three seasons after LMP establishment.

Hypothesis 6: GHG emissions will be higher in DTGW and DTGW+TD than in CT, due to the increased soil moisture from diversion terraces.

Hypothesis 7: Soil nitrate will be higher in DTGW and DTGW+TD than in CT, since nitrate leaching declines with declining soil water content.

Objective 3 is addressed in Chapter 3 of this thesis.

Study objective 4 was to quantify the GHG fluxes in response to volumetric water content

(20%, 27%, 35%, and 40% VWC), nitrogen fertilizer application rates (0 and 170 kg N ha⁻¹), and temperature (10 and 25°C) with a 6-day soil incubation experiment.

Hypothesis 8: CO₂ fluxes will decline at 35% VWC as this equates to 69% water-filled pore space (WFPS).

Hypothesis 9: N₂O fluxes will be highest with 35 and 40% VWC and 25°C

temperature. These VWC values equate to 69% and 79% WFPS. Nitrous oxide is

generally produced in soils with WFPS > 50%.

Hypothesis 10: CH₄ will begin to act as a source (i.e., positive fluxes) at a VWC of 40%, due to anaerobic conditions caused by displacement of air, by water, in pore spaces.

Objective 4 is addressed in Chapter 4 of this thesis.

Chapter 2: Short-term Response of Soil Aggregate Stability and Labile Carbon to Contour Tillage, Diversion Terrace, Grassed Waterway, and Tile Drainage Implementation

2.1 Introduction

Soil degradation has been of great concern for New Brunswick's potato farmers for several decades. With over 21,000 hectares under potato production, New Brunswick farmers are contributing 40% of Canada's potatoes (Statistics Canada, 2021). Potato production requires frequent tillage operations that inevitably break down aggregates and enhance organic matter decomposition (Grandy et al., 2002), in turn leading to enhanced erosion susceptibility (Le Bissonnais, 1988). Several studies have shown that soil organic matter (SOM) content in intensive potato production fields in New Brunswick has been declining since the 1960s (Eilers et al., 2010; Rees et al., 2014). As SOM levels decline, soil functioning is adversely affected, leading to reduction of biological activity and further degradation of soil structure (Haynes & Tregurtha, 1999; Gagnon et al., 2001).

Negative effects of intensive potato production are particularly noticeable in regions with shallow soils and hilly landscapes, as is the case in northwest New Brunswick (Langmaid et al., 1980; Saini & Grant, 1980; Coote et al., 1981; Wang et al., 1984; Edwards et al., 1998; Chow et al., 1999). Hence, many potato farmers in New Brunswick have begun using diversion terraces (DT) and grassed waterways (GW) as means to combat those negative effects associated with this intensive crop production system. While DT and GW have been extremely effective at reducing runoff and soil loss during a decade-long study in northwest New Brunswick outside the city of Grand Falls, these structures were not able to stop the decline of soil carbon (Rees et

al., 2008; Chow et al., 2010). This emphasized the need to combine several management practices to address all the negative issues associated with the intensive potato production.

In 2017, the Landscape Integrated Soil and Water Conservation (LISWC) was proposed by Dr. Sheng Li of the Fredericton Research and Development Centre (FRDC) as a new, holistic system designed to conserve soil and water on sloping landscapes by integrating multiple best management practices (BMPs) and studying their interactions (Li, 2021). Individual BMPs included in the LISWC system are diversion terraces (DT), grassed waterways (GW), tile drainage (TD), water retention structures, supplemental irrigation, contour tillage (CT), and soillandscape restoration. At present, only one study has been completed at the LISWC site. It focused on the removal of excess water from the tile-drained LISWC fields relative to undrained fields that otherwise had the same combination of BMPs (Dobson, 2022). This study found the tile-drained LISWC fields improved field-scale drying, relative to non-tile drained fields, within the first 48 hours after heavy rainfall events. This could have implications on soil quality indicators that are affected by changes in soil moisture.

No studies in the area have investigated the effects of these combined BMPs on aggregate stability, which is key in stabilizing soil and organic matter, regulating air and water infiltration, and protecting against erosion (Carter, 2002). Because soil aggregate stability generally correlates negatively with soil water content (Caron et al., 1992; Rasiah et al., 1992), the changes in hydrology when TD is added to DT and GW may affect aggregates and the organic matter occluded within. Generally, tile drainage may increase aggregate stability, especially of the largest aggregates (Abid & Lal, 2009; Kumar et al., 2014), but there have been no studies focused on the effects of combined BMPs on aggregate stability and SOM fractions. Typically, DT and GW studies in Atlantic Canada focused on soil hydrology or potato yield, with SOC as

the main indicator of soil health (Chow et al., 1990; Chow et al., 1999; Rees et al., 2008). None have investigated labile carbon (C) fractions, which are sensitive indicators of changes in total SOM (Carter, 2002). Permanganate oxidizable carbon (POXC) and particulate organic carbon (POC) are key labile C fractions that reflect processes such as soil aggregation and soil C accrual (Wardle, 1992; Six et al., 1998; Wander, 2004). A study in Missouri investigated the combination of terraces and underground tile outlets in cover and no-cover cropping systems on total organic C (TOC), labile C (POXC) and water stable aggregates – finding that two years after construction of the terraces, labile C, TOC, and water stable aggregates all declined (Adler et al., 2020).

The objective of this study was to evaluate effects of two land management practices (LMPs) (each comprised of several individual BMPs) on SOC, aggregate stability, permanganate oxidizable C (POXC), and particulate organic C (POC), relative to CT, across three slope positions in the second and third years after LMP implementation. The two LMPs included the following combinations of BMPs: (i) DT, GW, and CT [from now on labeled DTGW] and (ii) DT, GW, CT, and TD [from now on labeled DTGW+TD].

2.2 Materials and Methods

2.2.1 Site Description

The study was conducted between May 2020 and October 2021 on a field experimental site maintained by the FRDC of Agriculture and Agri-Food Canada in New Brunswick. The study was located on a northeast-facing slope facing the Saint John River (45°55'33.5"N 66°36'55.5"W). The whole Fredericton area, including the study site location, has been subject to glaciation, and most upland soils have been developed from weathered glacial till parent materials (Zelazny, 2007). The study site is located on soils that were part of the Research

Station Association and classified as Orthic Humo-Ferric Podzols. They developed in areas with loose ablation till over compact lodgment till (Rees & Fahmy, 1984). The soils are moderately permeable sandy loam with 10 to 25% angular cobbles and gravels, averaging 55% sand, 34% silt, and 10% clay (Appendix B), with a pH of 5-6. The Fredericton area receives a mean annual rainfall of 885.9 mm and a mean annual snowfall of 214.8 cm (Government of Canada, 2019).

Construction of the LISWC system on the study site began in 2018 and finished in the same year. Prior to construction, the site was under grass for two seasons. Before those two seasons, several strips were plowed along the slope for a rainfall simulation experiment, but nothing was planted. When construction began in 2018, old clay drainage tiles (dating to the mid-20th century) were found. The tiles were removed from the site after excavation. They were no longer functional and would have had marginal impact on soil moisture and water flow. In the initial 2019 season the study site was seeded to oat (*Avena sativa* L.), but this season was not factored into this study. In 2020, the site was again seeded to oat, while in 2021, a potato (*Solanum tuberosum* L.) crop was planted.

The study consisted of three land management practices (LMPs) constructed on a slope with a gradient of 10% (Figure 2-1). Each LMP took up a total of 1400 m² (70 × 20 m). Each LMP represents a different combination of individual BMPs. The 1st LMP acted as a control and consisted only of contour tillage – it will be referred to from here on as CT. The only permanent BMP installed in CT was the water retention structure at the base of the slope. There were 6 plots within CT.

The 2nd LMP was subdivided into 3 segments, each with 2 plots. Diversion terraces and grassed waterways were constructed below each of the three 20×20 m blocks. A water retention structure was constructed below the lowest block. This LMP is referred to as DTGW. The 3rd

LMP, DTGW+TD, was a repeat of DTGW but with tile drainage installed, which is representative of the full LISWC system. Two polyvinyl chloride (PVC) tile drains, measuring 10 cm in diameter, were installed per block, 10 m apart. Edges of the LMPs were lined with an impermeable plastic barrier extending to a depth of 1 m, to isolate each LMP from potential inflows from other LMPs. Contour tillage was implemented in both DTGW and DTGW+TD.



Figure 2-1 Schematic of the study site layout and the 3 land management practices (LMP) – CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage).

2.2.2 Field Seasons

2.2.2.1 2020 Field Season

Timeline of events
The experiment site was disked on June 5th (Appendix C), followed by additional disking and harrowing on June 15th, and oats were planted. On the same day as planting, a 17-17-17 urea-based fertilizer was applied at a rate of 337 kg ha⁻¹ (57 kg N ha⁻¹). An herbicide was sprayed on July 6th. Oats were harvested on September 22nd to prevent them from seeding in subsequent field seasons, the straw was mowed and left on the ground. No data was collected on oat yield. A final plowing with a moldboard plow took place on October 1st.

2.2.2.2 2021 Field Season

The CT management practice consisted of 68 total potato rows. In both DTGW and DTGW+TD, each plot had a total of 16 rows, for a total of 48 rows per LMP. Oats were grown at the base and top of CT, as well as at the bottom and top of each of the 3 plots in both DTGW and DTGW+TD to prevent soil from washing out at the flumes into the water retention pond.

Timeline of events

On June 7th Russet Burbank potatoes were planted with rows spaced at 0.91 m and plants spaced at 0.38 m within rows. A 17-17-17 urea-based fertilizer was applied during planting at a rate of 1120 kg ha⁻¹ (190.5 kg N ha⁻¹). This was slightly lower than the 208 kg N ha⁻¹ fertilizer rate that is the typical maximum fertilization rate for potato farming in New Brunswick (Government of New Brunswick, n.d. -d). Potato rows were hilled on July 20th. Insecticide was sprayed on both July 23rd and August 11th to suppress Colorado potato beetle (*Leptinotarsa decemlineata*) activity. Fungicide (Admire) spraying was scheduled every Friday (weather permitting) but sometimes took place a day earlier or later depending on rain events. Spray dates and a full timeline of events appear in Appendix D. Potato harvest took place on October 6th.

The 2021 field season was considerably wetter than 2020 and without as many peak temperatures in the mid-30°C. In June, July, and August of 2020, a total of 168 mm of

precipitation fell, whereas 258.4 mm fell in the same three months of 2021 (Government of Canada, 2021). The weather station on the study site recorded 51.2 mm or rain on July 9th during the remnants of Hurricane Elsa. Minor rill erosion was observed in all three land management practices the following day, causing exposed potato plant roots and small channels oriented up-and-down slope (Appendix E).

2.2.3 Sampling and Analysis

In 2020, composite soil samples were collected (Appendix C) on July 2nd and September 24th for all analyses except bulk density, which was done on August 19th (A and B subplots) and August 26th (C subplots).

In 2021, composite soil samples were collected on May 27th and October 7th. All samples from both years were air dried and passed through a 2-mm sieve. Each composite sample was comprised of 5 to 6 cores (0-15 cm depth) collected by Dutch auger using a random sampling strategy within each subplot.

Each soil sampling date had a total of three samples per plot, for a total of 54 (A, B, and C subplots) samples per date. The C subplots were added late and had to be located in the middle to match inventories and sampling patterns. For aggregate stability, POXC, and total C, all 54 samples were used. For time-consuming analyses like density fractionation of POC, only 36 samples were used (A and B subplots).

	СТ			DTGW			DTGW+TD		
1A	1C	1B	7A	7C	7B	13A	13C	13B	Upslope
2A	2C	2B	8A	8C	8B	14A	14C	14B	
3A	3C	3B	9A	9C	9B	15A	15C	15B	
4A	4C	4B	10A	10C	10B	16A	16C	16B	
5A	5C	5B	11A	11C	11B	17A	17C	17B	Downslope
5B	6C	6B	12A	12C	12B	18A	18C	18B	

Figure 2-2 Sampling locations for the 3 land management practices – CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage). Each LMP contained 3 blocks, which each contained 2 plots. Each plot was then further subdivided into 3 subplots containing the A, B, and C sampling locations.

2.2.3.1 Soil Organic Carbon

Soil organic carbon (SOC) concentrations were determined by dry combustion on 0.2-g subsamples using an Elementar varioMACRO apparatus (Elementar Americas Inc., Mt. Laurel, NJ).

2.2.3.2 Soil Aggregate Stability

Soil aggregate stability was determined using a 08.13 Wet Sieving Apparatus

(Eijkelkamp, 2008) on soil samples sieved to 1-2 mm collected on July 2nd, 2020 and October

7th, 2021. Results were determined according to modified Eijkelkamp operating instructions

(Wilson et al., 2018; Zebarth et al., 2022). Soil aggregate stability (SAS) is calculated using the

following equation:

Equation 2.1

$$SAS = ((Sa-Sw)/(T-Sw)) * 100$$

where *Sa* is the stable aggregate weight after H₂O (dry stable aggregate portion + sand portion), *Sw* is the dry sand weight, and *T* is the total dry weight (loose + stable + sand portions).

2.2.3.3 Permanganate Oxidizable Carbon

Permanganate oxidizable carbon was determined on samples collected from the following sampling dates: July 2nd and September 24th, 2020; and May 27th and October 7th, 2021. Results were analyzed according to Weil et al. (2003), with procedures described by Culman et al. (2012) (Appendix H).

Samples were analyzed at 550 nm using a Biochrom Libra spectrophotometer (Biochrom Ltd., Cambridge, United Kingdom) and calculated using the following equation:

Equation 2.2

 $POXC (mg kg^{-1}) = [0.02 mol/L - (a+b*Abs)] * (9000 mg C/mol) * (0.02 L solution/Wt)$ where 0.02 mol/L is the initial solution concentration, *a* is the intercept of the standard curve, *b* is the slope of the standard curve, *Abs* is the Absorbance of Unknown, 9000 is the milligrams of carbon oxidized by 1 mole of MnO₄ changing from Mn⁷⁺ to Mn²⁺, 0.02 L is the volume of stock solution reacted, and *Wt* is the weight of air-dried soil sample in kg.

2.2.3.4 Particulate Organic Carbon

Density fractionation used to determine particulate organic carbon was performed on soil samples collected on: July 2nd, 2020; September 24th, 2020; May 27th, 2021; and October 7th, 2021. The protocol was based on the method described in Poeplau et al. (2018) with one alteration – removal of the sieving at 0.2 mm (Appendix J). Hence, the protocol included the following three specific fractions: (i) fPOM (free light POM separated from oPOM); (ii) oPOM (sand-sized and occluded POM, 53-2000 μ m); and (iii) MAOM (silt- and clay-sized mineral associated organic matter, < 53 μ m rinsed from catch pan). Final samples were analyzed for total

C concentrations using an Elementar varioMACRO apparatus (Elementar Americas Inc., Mt. Laurel, NJ).

2.2.4 Statistical Analysis

A linear model was used to assess the *change* in SOC, aggregate stability, POXC, and the C concentration of the three fractionated pools (each fraction assessed individually). Change was assessed from July 2nd, 2020, to October 7th, 2021. Land management practice (CT, DTGW, and DTGW+TD) and slope position (upper, middle, and lower) were considered as fixed main effects. If there was no interaction between LMP and slope position, the interaction was dropped from the model. A Tukey post-hoc HSD test was performed using the *emmeans* function (Lenth, 2018) when ANOVA results showed significant differences between main effects ($\alpha = 0.05$). The normality and variance homogeneity assumptions of ANOVA were confirmed using Shapiro-Wilks test and Bartlett's test, respectively, in addition to residual diagnostic plots. Statistical analyses were conducted using R software version 4.0.3 (R Core Team, 2020).

2.3 Results

2.3.1 Soil Organic Carbon

Soil organic carbon did not change substantially from post-planting 2020 to post-harvest 2021 across LMP or slope position (Figure 2-3). At the study period beginning, total SOC in CT, DTGW, and DTGW+TD averaged 18.7 ± 0.5 , 18.5 ± 0.4 , and 16.3 ± 0.7 g C kg soil⁻¹, respectively. At the end of the study, SOC values averaged 18.0 ± 0.5 , 18.5 ± 0.4 , and 15.9 ± 0.8 g C kg soil⁻¹, respectively. There was approximately 21% more C in the lower slopes than in the upper or middle slopes. Across the entire study period, SOC in the upper, middle, and lower slopes averaged 16.4 ± 0.4 , 16.6 ± 0.3 , and 20.0 ± 0.3 g C kg soil⁻¹, respectively.

In the model assessing change in SOC through the length of the study period, LMP (p = 0.79) and LMP × slope position interaction (p = 0.50) was not significant at a statistical threshold of $\alpha = 0.5$ (Figure 2-4). Slope position (p = 0.06) was suggestive but inconclusive.



Figure 2-3 Soil organic carbon concentration within each land management practice (LMP) at post-planting 2020 and post-harvest 2021 for contour tillage - CT, diversion terraces and grassed waterways - DTGW, and diversion terraces, grassed waterways, and tile drainage - DTGW+TD, shaded by slope position. Error bars represent the standard error of the mean (n = 6).



Figure 2-4 Change in soil organic carbon (SOC, g C kg soil⁻¹) between post-planting 2020 and post-harvest 2021. Boxplots represent one slope position (upper, middle, or lower) per land management practice – CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage). Neither treatment nor slope position were significant ($\alpha = 0.05$). Boxplots display the median, 25th and 75th percentiles, whiskers, and individual outliers (dots), for this and all subsequent boxplots. Dashed red line equals 0 change.

2.3.2 Soil Aggregate Stability

No significant differences were found in aggregate stability among LMPs (p = 0.62) or

slope positions (p = 0.84), likely due to the large variability in terraced management practices

(Figure 2-5). There was no interaction between LMP and slope position (p = 0.61).

Mean aggregate stability percentages in CT, DTGW, and DTGW+TD were 92.1 \pm 0.4,

 93.2 ± 0.3 , and 88.6 ± 0.6 in 2020 and 87.7 ± 0.7 , 89.1 ± 0.6 , and 85.3 ± 1.1 in 2021,

respectively. Soil aggregate stability percentages across all LMPs were between 83.7 and 95.9%

in July 2020 and from between 77.53 and 97.07% in October 2021.

Mean aggregate stability percentages in upper, middle, and lower slopes were 91.6 ± 0.7 , 90.2 ± 0.8 , and 92.1 ± 0.4 in 2020 and 87.7 ± 0.6 , 86.6 ± 0.8 , and 87.9 ± 1.1 in 2021, respectively.

The only LMP that showed any increase in aggregate stability was DTGW+TD. However, only 3 of 18 samples showed an increase: 1 from the middle slope position and 2 from the lower slope. The most variability overall was seen in DTGW+TD, while CT had the least variability. Although aggregate stability declined in all LMPs, the measured aggregate stability values were all very high (Appendix G).



Figure 2-5 Percentage change (%) in soil aggregate stability (SAS) between post-planting 2020 and post-harvest 2021 in CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage).

2.3.3 Permanganate Oxidizable Carbon

Clear trends across LMPs were evident with POXC data. In CT, POXC values declined over the sampling period (Figure 2-6). In CT, average POXC values began at 319.71 ± 8.5 mg kg⁻¹ soil on July 2nd, 2020 and measured 282.2 ± 9.1 mg kg⁻¹ soil on October 7th, 2021. Meanwhile, in the terraced LMPs, POXC trended upward over time. In DTGW and DTGW+TD, POXC values measured 265.7 ± 8.3 and 191 ± 12.1 mg kg⁻¹ soil, respectively, on July 2nd, 2020, and measured 312.6 ± 9.8 and 278.3 ± 11.7 mg kg⁻¹ soil, respectively, on October 7th, 2021.

As shown in Figure 2-7, CT mostly lost labile C over the 16-month study period, averaging a loss of 11.2% across LMP and slope. It is possible that C in this form washed away each growing season. Terraced LMPs, on the other hand, gained POXC. The more that water was regulated, the more POXC accumulated; DTGW gained an average of 19.8%, while DTGW+TD rose an average of 50.6%.

Land management practice (p < 0.001) and the interaction between LMP and slope position (p < 0.01) were significant in the model, while slope position itself was not (p = 0.19).



Figure 2-6 Permanganate oxidizable carbon (POXC) values in CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage). Soil measurements were taken on July 2nd and September 24th, 2020 and May 27th and October 7th, 2021. Boxplots represent one slope position (upper, middle, or lower) per LMP.



Figure 2-7 Percentage change (%) in permanganate oxidizable carbon (POXC) between post-planting 2020 and post-harvest 2021 in CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage). There was a significant LMP × slope position interaction ($\alpha = 0.05$).

2.3.4 Particulate Organic Carbon

The concentrations of C in each LMP changed over the study period (Figure 2-8). Overall, CT lost an average of 0.58 ± 1.4 g C kg⁻¹ soil (-3.75%) in the MAOM fraction, while DTGW+TD gained an average 1.7 ± 0.9 g C kg⁻¹ soil (+9.7%). The biggest change was seen in DTGW, where C in the MAOM fraction rose by an average of 2.5 ± 1.03 g C kg⁻¹ soil, or roughly +15.8%. There was considerably more C in the MAOM fraction than in other fractions. Concentrations, across all LMPs, slope positions, and dates, averaged 2.0 ± 0.04 , 1.65 ± 0.1 , and 13.6 ± 0.3 g C kg soil⁻¹ in fPOM, oPOM, and MAOM, respectively. What is interesting about the gains made in terraced LMPs is that they occurred while the whole mass of the MAOM fractions recovered declined (Figure 2-9). An average decline of 4.85% to 417 g kg soil⁻¹ occurred in DTGW, while DTGW+TD declined an average of 10.6% to 415 g kg soil⁻¹. There was an increase of MAOM C in 75% of DTGW subplots and in 66% of DTGW+TD subplots, particularly in the middle and lower plots.

The fPOM C concentration trended slightly lower with increasing LMP water management (i.e., CT>DTGW>DTGW+TD), though differences were minimal. The fPOM concentrations in the CT treatment had an average 2.14 ± 0.1 , while DTGW had an average 2.07 ± 0.1 , and DTGW+TD had an average 1.89 ± 0.1 g C kg⁻¹ soil. However, CT lost the most g C kg⁻¹ soil over the 16 months of this study (-0.34 ± 0.1), while DTGW+TD lost the least (-0.09 ± 0.1) (Figure 2-5).

The oPOM trends were very similar to those of fPOM. Contour tillage had the highest average g C kg⁻¹ soil throughout the two years (1.97 ± 0.1) but lost the most C over the 16 months (-0.32 ± 0.1), whereas DTGW+TD had the lowest average (1.39 ± 0.04) but gained C (+ 0.22 ± 0.1).



Figure 2-8 Organic C concentrations per kg of soil between post-planting 2020 and post-harvest 2021 in CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage).



Figure 2-9 Top: Change in organic C concentration per kg of soil between post-planting 2020 and post-harvest 2021 in CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage). Bottom: Percentage change in the mass of each fraction between the same dates.

2.4 Discussion

2.4.1 Soil Organic Carbon

Soil organic carbon (SOC) was not expected to change significantly over the course of the second and third years after LMP establishment. The turnover of SOC is generally slow, even in cases where intensive potato production takes place. In the Chow et al. (2010) study on impacts of long-term intensive potato production and terraces on runoff hydrology and soil quality, for instance, SOC only declined 3.6% over the course of an entire decade. Once a new management practice is implemented, measurable differences in SOC arising from altered soil management can often take years to be detected and may not be picked up in a short-term study (Sikora et al., 1996). This is the impetus for many of the methods that quantify labile soil C rapidly and inexpensively, such as particulate organic carbon and POXC, that are included in this study.

2.4.2 Soil Aggregate Stability

In the second and third year after LMP establishment, the CT management practice showed a small but clear trend where aggregate stability decreased more in the elevated slope position and less in the lowest slope position (Figure 2-5). The opposite trend appeared in the terraced LMPs, where aggregate stability decreased more on lower slopes.

It is not surprising that soils, which were grass pasture before site construction, showed high aggregate stability values above 80%. The LISWC system had only been built in 2018, and no potato cultivation had taken place there in the past decade, so the soil had become well-aggregated. The oat crop from 2020 also required very few tillage events.

Also, 2020 was warmer and drier than typical for this region, which could have further contributed to better aggregate stability. Soil aggregate stability is generally negatively correlated

with soil water content, since aggregates tend to become stronger with drying (Caron et al., 1992; Rasiah et al., 1992). Percentages in the 80 to 90% range are on the higher end for potato cropping systems in the area. A study on similar Fredericton-area soils with compost amendments to potato crops had values as low as 67.7% (Wilson et al., 2018), and a biocharamended soil at the FRDC in 2015 reported values in the 70% range (Delisle, 2018). Aggregate stability averaged 68% in a province-wide study of commercial fields between 2013 and 2017, with values as low 32% (Zebarth et al., 2022). Longer term monitoring of this site would reveal whether or not diversion terraces could assist in retaining high aggregate stability relative to contour tillage with repeated potato cropping.

Other research in Atlantic Canada has demonstrated that certain BMPs can increase water-stable soil aggregates either through conservation tillage (Carter & Sanderson, 2001; Carter et al., 2009) or crop rotation (Angers et al., 1999), but those were 6-year studies. There have been few studies in the area specifically focused on terraces and/or tile drainage effects on aggregate stability. There is evidence elsewhere that tile drainage can increase macroaggregation but negatively affect other aggregate size classes, as seen in a study comparing tile-drained and non-drained Ohio soils (Kumar et al., 2014). Increasing water-stable aggregates is a slow process (Rees et al., 2014), and it is possible that over a longer study period that patterns may emerge, given the differences in variation between the three land management practices.

2.4.3 Permanganate Oxidizable Carbon

Soil organic matter was determined directly through changes in SOC and through changes in POXC and POC. These labile SOM fractions – POXC and POC – have been identified in other studies to be sensitive to management practices (Bongiorno et al., 2019; Culman et al., 2012; Haynes, 2005; Hurisso et al., 2016). By monitoring trends in these pools of

carbon, it is possible to get an earlier indication of the consequences of different land management practices than monitoring total organic matter (Dalal & Mayer, 1986a,b). Little is known about how SOC pools respond to the effects of soil water redistribution from the implementation of diversion terraces, grassed waterways, and tile drainage.

In each of the two study years, the following trends were evident between planting and post-harvest - a decrease in labile POXC in CT and an increase in the terraced LMPs. However, between fall of 2020 and spring of 2021, the trend in all three LMPs reversed – but did not completely revert to starting values – before resuming their respective trends (Figure 2-6). This suggests a seasonal effect on POXC. Culman et al. (2012) found POXC to be significantly related to microbial biomass C and closely related to smaller-sized (53–250 µm) particulate organic carbon (POC) fractions. As microbial biomass C might decline in the freezing conditions of a Maritime winter, and small POC fractions should be washed away in the spring snow melt, I would expect that POXC would decrease in all LMPs over winter - perhaps more in CT where there are no terraces to slow water runoff. But this was not the case in CT, where it increased over winter. One potential explanation might be that the presence of diversion terraces increased the soil moisture heading into winter and therefore also decreased the soil temperature. This could slow down the microbial processing of this carbon fraction. However, tile drainage has been shown to increase soil temperature (Fraser & Fleming, 2001), and since there was no difference in POXC trends between DTGW and DTGW+TD over winter, soil temperature is not likely the reason POXC values went up over winter in CT.

An Ohio study that compared tile-drained and undrained cropland found that the loss of readily mineralizable C was roughly 4 times greater in tile-drained land (Jacinthe et al., 2001). Not only did the tile-drained soil in our study gain labile C instead, but the tile-drained terraced

soil gained more labile C than the undrained terraced LMP. Other studies also document higher depletion rates of labile C pools when native ecosystems are converted to agriculture. (Elliott, 1986; Carter et al., 1998).

Studies have shown a close, positive relationship between POXC and SOC, and POXC is considered to be a relatively processed fraction of labile C (Culman et al., 2012). Study by Culman et al. (2012) specifically showed the fraction was more closely related to heavier than lighter particulate organic C (POC) and could thus be considered a good indicator of organic matter stabilization. The POXC trends in land uses were in line with the heavier POC pools fractioned in this study (oPOM, MAOM). Short-term changes in POXC reflect long-term changes in SOM (Weil et al., 2003), and though total C was statistically unchanged between years in any LMP, these results indicate an increase in C stabilization in terraced LMPs over fields managed with contour tillage.

2.4.4 Particulate Organic Carbon

The majority of C (~78.7%) in my study was found in the MAOM fraction of the soil – very similar to the lower end of results from a study by Gregorich et al. (2009) on long-term tillage and crop rotations in Prince Edward Island, Québec, and Ontario.

With the C increasing in both terraced LMPs over the two-season study period, it is plausible to expect that those LMPs might eventually hold more C in the MAOM fraction if the trend continued. This is because, as Haddix et al. (2020) point out, the MAOM pool forms quickly from any labile C in the soil and, in soils with relatively low C content (< 3%), as in the soil in our study, there is positive feedback between new SOM stabilization and soil C. Both terraced LMPs saw large gaps between the gains made in upper and lower plots. It is possible there was some downslope movement of C in terraced LMPs, but it is unlikely for several

reasons. First, diversion terraces have proven effective at preventing erosion (Chow et al., 2010), so it is not likely that POM C or MAOM C were translocated downhill by way of soil erosion. The POM C changes were also minimal and would not be large enough to account for the change. Although dissolved organic carbon (DOC) may have moved through the soil profile, it generally makes up less than 2% of SOM (von Lützow et al., 2007). The DOC (data not shown) was generally equal in each LMP and slowly (and equally) declined throughout the study period across all treatments.

The MAOM fraction represents a more stable form of soil C due to its association with soil mineral particles and consequent protection from decomposition. From the perspective of climate change mitigation and soil health on potato fields in Atlantic Canada, it might be advantageous that MAOM increases over POM. Recently, Lavallee et al. (2020) proposed a simplified framework to assess SOM with only two components, with fundamental differences in formation, persistence, and function - particulate organic matter (POM) and mineral-associated organic matter (MAOM). The defining difference between the two being that POM is physically unprotected from decomposition processes, while MAOM is protected through association with silt- and clay-sized soil mineral particles. Such differentiation does not hold for my study, since recent results from numerous New Brunswick locations (unpublished data), including agricultural sites, have shown that the oPOM fraction contains the oldest C (L.P. Comeau, personal communication, March 30, 2022). If the oPOM C is able to persist the longest in New Brunswick agricultural soils, increasing its relative proportion would be more conducive to storing C long term. The fPOM fraction is not protected by organo-mineral complexes and is therefore a readily available substrate and nutrient source for micro-organisms (Gregorich et al., 1994). This fraction will turn over in Canada in 1-8 years (Carter, 2002) and is still important to

have in the soil so that SOC consistently turns over, stimulating plant productivity and releasing nutrients.

The changes in POM (fPOM + oPOM) C were small, given that the expectation was that with cultivation comes a decrease in POM. The persistence of POM is controlled mainly by microbial and enzymatic inhibition (Lavallee et al., 2020). Due to fPOM being less protected from the activity of microbes and enzymes, environmental changes that may decrease microbial inhibition or disrupt aggregates (e.g., tillage) can immediately increase the POM decomposition rates.

Because oats were seeded in 2020, I was not able to observe any follow-up effect of the 2021 potato crop on POM during my study period. The oats were harvested, but the straw was left to be incorporated into the soil in the fall. There is potential that in a season following potato, POM C (particularly fPOM C) might decrease significantly more, since so little organic matter inputs occur following potato harvest.

2.5 Conclusions

This study examined effects of three LMPs on indices of soil quality. It was hypothesized that aggregate stability would decrease the most in terraced LMPs, but there was no difference in the 2nd and 3rd seasons after LMP establishment. Because the diversion terrace and grassed waterway system diverts more of the surface water downwards into the soil profile, this was the rationale for the hypothesis. Perhaps soil was not wet enough in either growing season to induce greater loss in aggregate stability in terraced treatments over CT, or perhaps the effects of higher soil water content were masked in the short term when intensive tillage was taking place.

Contour tillage (CT) averaged a loss of 11.2% POXC, while terraced LMPs – DTGW and DTGW+TD – gained 19.8% and 50.6%, respectively. The more water was regulated, the higher

labile C rose. The data from this study show that, at least in the short term, labile C can rise shortly after the construction of BMPs that include diversion terraces and grassed waterways, with or without tile drainage.

There was an increase in MAOM C in the terraced LMPs, despite the high amount of tillage events that took place during potato cultivation. I also saw an increase in MAOM C concentrations in the terraced LMPs despite the mass of the whole fraction in each LMP declining.

Overall, there is evidence that the LISWC system has positive impacts on the soil properties evaluated. Though both terraced LMPs saw a decline in aggregate stability, neither were any worse than the CT. The diversion terraces and grassed waterways increased the fraction of POXC relative to CT. The further addition of tile drainage contributed to even higher POXC increases. Diversion terraces and grassed waterways also increased the fraction of organic matter associated with mineral particles, which represents a type of carbon that tends to have a slower turnover rate. This indicates that carbon is being stabilized and SOC may begin to increase due to the change in soil management. Chapter 3: Greenhouse Gas Emissions and Nitrate Supply Rate Response to Initial Three Years of Diversion Terrace, Grassed Waterway, and Tile Drainage Implementation

3.1 Introduction

The Landscape Integrated Soil and Water Conservation (LISWC) is a new system, designed five years ago, to conserve soil and water resources on sloping landscapes by integrating multiple BMPs (Li, 2021). The system combines diversion terraces, grassed waterways, tile drainage, water retention structures, supplemental irrigation, conservation tillage practices, and soil-landscape restoration. There are concerns, however, that in wet years the diversion terraces can lead to fields being too wet (Chow et al., 2010) since diversion terraces slow down surface drainage and direct more water into the soil profile. In turn, this can also lead to cooler soil temperatures. Soil water and temperature are the two most influential parameters for soil gas emissions (Oertel et al., 2016). Soil moisture controls microbial activity and related microbial processes by solubilizing substrate compounds and, through diffusion, increases substrate availability to active microbial sites (Sierra et al., 2017). Heightened microbial activity will then lead to higher decomposition of organic matter and higher microbial respiration rates that produce GHGs in soils. Rising soil temperatures also increase microbial metabolic rates, increasing respiration further (Larcher, 2003). With too much water in the system (i.e., saturated soil), anaerobic conditions foster the production of CH₄ and N₂O from methanogenesis and nitrification/denitrification, respectively. The dominant process in humid climates such as in New Brunswick is denitrification (Mosier, 1998).

Crop management practices can affect GHG emissions. Destructive management like the frequent tillage in potato farming can break down aggregates and expose organic matter to oxidation or decomposition processes. In general, carbon (C) sequestration will be more successful in land management systems that minimize erosion and soil disturbances, maximize return of crop residues, or maximize the efficiency of crop water and nutrient use (Paustian et al., 2004). Because of the intensive nature of most cropping systems – particularly potato production systems – we could mitigate GHG production and store more C and nitrogen (N) in the soil by modifying soil management practices (Smith et al., 2008).

Because nitrate (NO_3^-) is a soluble substance prone to leaching, the water redistribution from diversion terraces, grassed waterways, and tile drainage may affect the NO_3^- concentrations within the soil. Emissions of N_2O are a direct function of nitrogen (N) application rate, and the NO_3^- supply rate within the soil to plants, after fertilization, may be affected by the different management practices studied – contour tillage (CT), diversion terraces and grassed waterways (DTGW); and diversion terraces, grassed waterways, and tile drainage (DTGW+TD).

The objective of this study was to evaluate the effects of diversion terraces, grassed waterways, and tile drainage, relative to contour tillage, on GHG emissions from the soil and on nitrate supply rates that may relate to N_2O production during the initial three years of management implementation. Information gathered will help understand how GHG fluxes respond to altered water regimes under the differing land management practices. There have been no studies in the area that quantify GHG emissions from fields with diversion terraces and grassed waterways.

3.2 Materials and Methods

3.2.1 Site Description

The field experiment was conducted between May 2020 and October 2021, at the Fredericton Research and Development Centre (FRDC) of Agriculture and Agri-Food Canada. The FRDC is located on the south-east edge of the city of Fredericton, New Brunswick, while the study site was located on a northeast facing slope facing the Saint John River (45°55'33.5"N 66°36'55.5"W). The topography of the Fredericton area ranges from 6 to 215 m above mean sea level. The whole area has been subject to glaciation and most upland soils have been developed from weathered glacial till parent materials (Zelazny, 2007). The soils were part of the Research Station Association, classified as Orthic Humo-Ferric Podzols, and developed in areas with loose ablation till over compact lodgment till (Rees & Fahmy, 1984). They are moderately permeable sandy loam with 10 to 25% angular cobbles and gravels and a pH of 5-6. The soil texture is classified as a sandy loam, averaging 55% sand, 34% silt, and 10% clay (Appendix B). The Fredericton area receives a mean annual rainfall of 885.9 mm and a mean annual snowfall of 214.8 cm (Government of Canada, 2019).

Construction of the LISWC system began in 2018. Prior to construction, the site was under grass for two seasons. Before those two seasons, several strips were plowed along the slope for a rainfall simulation experiment, but nothing was planted. When construction began in 2018, old clay drainage tiles (dating to the mid-20th century) were found. The tiles were removed from the site after excavation. They were no longer functional and would have had marginal impact on soil moisture and water flow. In the initial 2019 season the study site was seeded to oat (*Avena sativa* L.). In 2020, the site was again seeded to oat, while in 2021 a potato (*Solanum tuberosum* L.) crop was planted.

The study consisted of three land management practices (LMPs) constructed on a slope with a gradient of 10% (Figure 2-1). Each LMP took up a total of 1400 m² (70 × 20 m). Each LMP represents a different combination of individual best management practices. The 1st LMP acted as a control and consisted only of contour tillage on each year's crop – it will be referred to from here on as CT. The only permanent BMP installed in CT was the water retention structure at the base of the slope. There were 6 plots within CT.

The 2nd LMP was subdivided into 3 segments, each with 2 plots. Diversion terraces and grassed waterways were constructed below each of the three 20 × 20 m blocks. A water retention structure was constructed below the lowest block. This LMP is referred to as DTGW. The 3rd LMP, DTGW+TD, was a repeat of DTGW but with tile drainage installed, which is representative of the full LISWC system. Two polyvinyl chloride (PVC) tile drains, measuring 10 cm in diameter, were installed per block, 10 m apart. Edges of the LMPs were lined with an impermeable plastic barrier extending to a depth of 1 m, to isolate each LMP from potential inflows from other LMPs. Contour tillage was implemented in both DTGW and DTGW+TD.



Figure 3-1 Schematic of the study site layout and the 3 land management practices (LMP) – CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage).

3.2.2 Sampling and Analysis

3.2.2.1 CO₂ Readings

To sample CO₂ fluxes, an EGM-5 Portable CO₂ Gas Analyzer (PP Systems, Amesbury,

MA, USA) with an SRC-2 Soil Respiration Chamber was used throughout the growing seasons

of 2020 and 2021. A HydraProbe sensor (Stevens Water Monitoring Systems, Portland, OR,

USA) was used with the EGM gas analyzer to gather surface soil moisture and temperature data

with each flux measurement.

Flux measurements were taken twice a week, weather dependent, and began at 10:00 am to coincide with daily mean soil temperatures. Fluxes were measured either for 60 seconds or until the threshold for change in CO₂ concentration was reached. Measurements in 2020 went from May 28th to September 22nd. Measurements in 2021 were able to begin earlier, ranging from April 29th to September 28th. Once collars for N₂O and CH₄ were installed in the field, measurements were taken inside the collars, which stayed free of oat or potato.

The linear respiration rate was calculated by the EGM with the following formula:

$$R = \frac{(Cn - C_0)}{Tn} * \frac{V}{A}$$

where *R* is the respiration/assimilation rate (CO₂ flux, or moles of CO₂ unit area⁻¹ unit time⁻¹), C_0 is the CO₂ concentration at T=0, and *Cn* is the concentration at finishing time *Tn*. The area of soil exposed is represented by *A*, and *V* is the total system volume (PP Systems, 2018).

3.2.2.2 N₂O and CH₄ Readings

Closed static dark chambers were used to make measurements of CH₄ and N₂O in the growing seasons of 2019, 2020, and 2021. A total of 36, cylindrical collars, measuring 38 cm tall and 30 cm in diameter were installed in the field at each subplot, at least 72 hours before the first measurement of the season (Appendix K). In 2019 and 2020, the site was seeded to oat, and any visible plant matter inside the collar was removed immediately after installation. In 2021, the collars were installed directly into potato rows on July 21st and 22nd. A single potato plant was pulled where each collar was to be installed, to ensure no seed potato or potato plant was inside the collar. This restricted the emissions generated from plant root respiration. In 2019, measurements were taken on August 28th and September 10th and 26th. In 2020, measurements

were taken on July 29th, August 12th and 26th, and September 4th. In 2021, measurements were taken on July 28th, August 10th and 25th, and September 8th and 21st.

On each day of measurement, plastic lids 30 cm in diameter were secured to the top of the collar with a tube at a diameter small enough to ensure the collar would be an air-tight, closed system. Lids from plastic, 5-gallon buckets were custom fit with 2 one-touch fittings (SMC Corporation, Tokyo, Japan). Each fitting had placed inside of it a ten-inch length of 6mm tubing with a one-way stopcock (Cole-Parmer, Montréal, Canada) at the halfway point. Stopcocks were used to ensure venting, so that buildup of any gas was prevented before the time 0 measurement. Lids also had a rubber septum in the middle where the syringe would draw a 20 mL gas sample into a pre-evacuated 12 mL exetainer.

Once all collars were secured with lids, measuring could begin. Four measurements were made at time 0, 30, 60, and 90 minutes. Measurements were staggered by 60 seconds so that there was time to draw a syringe sample and move to the next sampling location. Two people each took 18 samples at each of the 4 intervals, leaving 12 minutes between the end of one interval and the beginning of the next.

All gas samples from exetainers were analyzed at the University of British Columbia using a 7890A gas chromatograph (Agilent Technologies Inc., CA, USA) equipped with a flame ionization detector and electron capture detector. It was also equipped with a PAL auto-sampler (Agilent Technologies Inc., CA, USA). Fluxes were calculated with R, version 4.0.3, in R Studio with the *gasfluxes* package (Fuss et al., 2020). The *lin.fit* method from the package was used to fit a linear model via R's *lm* function, using concentration and time data.

Continuous soil moisture and temperature data were taken from the 5TE sensors (METER Group). Data was taken from the 11:00 am reading – halfway between start and end of

gas sampling – and from the 15 cm depth reading. During this study, measurements were only taken on bare soil – intentionally excluding plant root respiration and focusing solely on heterotrophic respiration.

3.2.2.3 Nitrate Supply Rate

The availability of nitrate was determined using Plant Root Simulator (PRSTM-Probes) technology from Western Ag Innovations Inc., Saskatoon (SK), Canada. The probes are ion exchange membranes (adsorbing surface area of 17.5 cm²/probe) encapsulated in a thin plastic probe. They were inserted into the 0–10 cm layer of the mineral soil where they adsorbed nutrients in their available forms. The probes also adsorbed ammonium ions, but a majority of the results were below method detection limits and were therefore not included. Probes were placed adjacent to the greenhouse gas sampling chambers in each of 2020 and 2021. In 2021, because the fields were seeded with potato, probes were only inserted into potato rows and not furrows. Probes were sent back to Western Ag for analysis after the removal of each round of probes.

The four sampling periods in 2020 were: (1) June 17 – July 12; (2) July 2 – July 28; (3) July 28 – August 18; and (4) August 19 – September 8. The three sampling periods in 2021 were: (1) July 26 – August 16; (2) August 16 – September 7; and (3) September 7 – September 22.

3.2.3 Statistical Analysis

The differences in GHG fluxes were analyzed using ANOVA and Tukey post-hoc tests. A linear mixed effects model was used with LMP (CT, DTGW, DTGW+TD) and slope position (upper, middle, lower) as fixed effects, subplot as a random effect to account for repeated measurements, and soil water content and temperature as continuous explanatory variables. A

model was fit for each gas separately. Soil temperature was not included in the N₂O model, as it was not significant in the model. A linear mixed effects model was also used to assess volumetric water content (VWC); only LMP, slope position, and their interaction were included in the model as fixed effects. For CO₂, soil moisture and temperature values from the HydraProbe sensor were included as continuous explanatory variables. A linear mixed effects model was fit with LMP to assess total greenhouse gases emitted as CO₂ equivalents, with 298 and 25 used as the conversion factors for N₂O and CH₄, respectively (IPCC, 2019). Finally, for nitrate supply rate, a linear mixed effects model was fit with LMP and slope position as fixed effects and subplot as a random effect.

Type III ANOVA was used to evaluate significance of interaction terms in the models. A Tukey post-hoc HSD test was performed using the *emmeans* function (Lenth, 2022) if the main effects LMP or slope position were significant without a significant interaction between the two. The ANOVA assumptions of normality and homogeneity of variance were assessed using Shapiro-Wilks test and Bartlett's test, respectively, in addition to diagnostic plots. Data transformations were made when necessary to ensure conditions of ANOVA were met. A statistical significance threshold of $\alpha = 0.05$ was used for hypothesis testing. Statistical analyses were conducted using R software version 4.0.3 (R Core Team 2020).

3.3 Results

3.3.1 Carbon Dioxide

Soil temperature, soil water, and LMP were all significant in each year. Slope position was not significant in either year, nor was the interaction between LMP and slope position. Figures 3-5 and 3-6 show the CO_2 fluxes from 2020 and 2021.

Mean daily emissions (kg CO₂-C ha⁻¹) in 2020 were 56.6 \pm 3.6, 68.8 \pm 4.8, and 56.7 \pm 3.7 in CT, DTGW, and DTGW+TD, respectively (Figure 3-7). In 2021 they averaged 30.1 \pm 1.4, 42.2 \pm 4.5, and 28.8 \pm 1.2. Daily emissions across the entire study period averaged 43.1 \pm 2.0, 55.3 \pm 3.3, and 42.5 \pm 2.0.

There were significant differences in total CO₂ emissions in both 2020 (p = 0.018) and 2021 (p < 0.001) (Figure 3-5). In 2020, total C emitted from CO₂ was 1640 ± 82.5, 1996 ± 112, and 1644 ± 74.7 kg C ha⁻¹, while in 2021 total C emitted was 903 ± 28.5, 1265 ± 105, and 863 ± 26 kg C ha⁻¹ in CT, DTGW, and DTGW+TD, respectively.

Emissions were correlated to soil temperature slightly more than soil moisture. Emissions were positively correlated to soil temperature at 15 cm in 2020 (r = 0.27, p < 0.001, n = 803) and 2021 (r = 0.10, p < 0.05, n = 468). In 2020, there was no correlation between emissions and soil moisture, while in 2021 fluxes were only weakly correlated to VWC (r = 0.10, p < 0.05, n = 468).

In 2020, VWC averaged 0.158 ± 0.003 , 0.169 ± 0.003 , and $0.200 \pm 0.005 \text{ m}^3 \text{ m}^{-3}$, while in 2021, VWC averaged 0.282 ± 0.005 , 0.313 ± 0.005 , and 0.277 ± 0.001 in CT, DTGW, and DTGW+TD, respectively. In 2020, soil temperature averaged 22.9 ± 0.3 , 23.2 ± 0.3 , and $24.1 \pm 0.3^{\circ}$ C, while in 2021, temperature averaged 21.6 ± 0.2 , 21.2 ± 0.2 , and $21.8 \pm 0.2^{\circ}$ C in CT, DTGW, and DTGW+TD, respectively.



Figure 3-2 A) CO₂ fluxes from May 28th to September 22nd, 2020 in the 3 land management practices: CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage). B) Mean daily temperatures and daily precipitation levels.



Figure 3-3 A) CO₂ fluxes from April 29th to September 28th, 2021 in the 3 land management practices: CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage). B) Mean daily temperatures and daily precipitation levels.



Figure 3-4 Mean daily emissions (kg CO₂-C ha⁻¹) for each LMP – CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage) – in each of the 2020 and 2021 seasons. Error bars represent the standard error of the mean (2020, n = 348; 2021, n = 360). Different letters indicate a statistically significant difference between LMPs using Tukey's HSD ($\alpha = 0.05$).



Figure 3-5 Total CO₂ emissions (kg C ha⁻¹) in 2020 and 2021 in each LMP – CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage). Error bars represent the standard error of the mean (n = 12). Different letters indicate a statistically significant difference between LMPs ($\alpha = 0.05$).



Figure 3-6 Soil temperature data in 2020 and 2021 for each LMP – CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage).



Figure 3-7 Soil volumetric water content data in 2020 and 2021 for each LMP – CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage).

3.3.2 Nitrous Oxide

Land management practice (p < 0.01), slope position (p < 0.05), and soil moisture (p < 0.001) were all significant in the model for N₂O. There was also a significant interaction (p < 0.01) between LMP and slope position.

In the VWC model, LMP was not significant (p = 0.71). However, slope position was significant (p < 0.01), and the lower slope position was significantly different from the mid (p < 0.01) and upper (p < 0.001) positions. N₂O was positively correlated with soil moisture (r = 0.23, p < 0.001) across all three seasons but was not, however, significantly correlated with soil temperature (r = -0.01, p = 0.74).

The N₂O fluxes in DTGW were higher than in CT or DTGW+TD (Figure 3-2). The mean daily efflux (kg N₂O-N ha ⁻¹) for CT, DTGW, and DTGW+TD across all 3 years was 0.0124 \pm 0.002, 0.015 \pm 0.0019, and 0.011 \pm 0.0016, respectively (Figure 3-3). Mean VWC measurements (m³ m⁻³) in CT, DTGW, and DTGW+TD were 0.23 \pm 0.01, 0.25 \pm 0.01, and 0.24 \pm 0.01, respectively.

The only significant difference between LMPs in total CO₂e emission of N₂O was seen in 2019 (p < 0.001) (Figure 3-11).

Mean daily effluxes from upper to lower slope positions were 0.0151 ± 0.002 , 0.0130 ± 0.002 , and 0.0106 ± 0.001 kg N₂O-N ha⁻¹ day⁻¹, respectively. The upper slopes were surprisingly the wettest and the lower slopes the driest. Soil VWC averaged 0.26 ± 0.01 , 0.25 ± 0.01 , and 0.22 ± 0.01 m³ m⁻³ top to bottom. The CT emissions increased from top to bottom. In terraced LMPs emissions increased bottom to top.

3.3.3 Methane

Mean CH₄-C oxidation rates across all years for CT, DTGW, and DTGW+TD were 1.17 ± 0.2 , 0.42 ± 0.5 , and 1.56 ± 0.2 g ha⁻¹ day⁻¹, respectively. Methane oxidation rates were positively correlated with soil moisture (r = 0.26, p < 0.001, n = 425) and negatively correlated with soil temperature (r = -0.24, p < 0.001, n = 425). There was a significant interaction between LMP and slope position (p = 0.048). Soil temperature and moisture were both significant in the
model (p < 0.01). There were no significant differences between LMPs seen in 2019 (p = 0.08), 2020 (p = 0.86) or 2021 (p = 0.17) (Figure 3-12).

The highest CH₄ uptake by slope position (-1.6 \pm 0.2 g ha⁻¹ day⁻¹) occurred in the lower slopes, which were ~16% drier than the upper slopes. This is in line with expectations for the driest soils. The upper and middle slopes oxidized 0.9 \pm 0.2 and 0.7 \pm 0.5 g ha⁻¹ day⁻¹.



Figure 3-8 Mean daily emissions of $N_2O(A)$ and $CH_4(B)$ in 2019, 2020, and 2021, with soil temperature (C) and soil moisture (D), and coloured by LMP – CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage).



Figure 3-9 Effect of land management practice (LMP) – CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage) – on emission of N_2O (A) and CH₄ (B) across 2019, 2020, and 2021 growing seasons. The bars of the columns represent standard errors of the mean.



Figure 3-10 Emission of N_2O by slope position in each land management practice – CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage).



Figure 3-11 Total N₂O emission for the 2019, 2020, and 2021 seasons expressed in CO₂ equivalents (CO₂e) (kg C ha⁻¹) for each LMP – CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage). Error bars represent the standard error of the mean (n = 12). Different letters indicate a statistically significant difference between LMPs ($\alpha = 0.05$), but comparisons can only be made within year.



Figure 3-12 Total CH₄ emission for the 2019, 2020, and 2021 seasons expressed in CO₂ equivalents (CO₂e) (kg C ha⁻¹) for each LMP – CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage). Error bars represent the standard error of the mean (n = 12). Different letters indicate a statistically significant difference between LMPs ($\alpha = 0.05$), but comparisons can only be made within year.

3.3.4 Nitrate Supply Rate

In 2020, there was a statistically significant interaction between LMP and slope position (p = 0.017). In 2021, not one of LMP (0.65), slope position (0.33), or their interaction (0.41) was significant.

In 2020, the highest supply rates were seen in DTGW between July 2^{nd} and July 28^{th} at $404 \pm 50.5 \ \mu g/cm^2$ (Figure 3-13). Rates in the July 26^{th} to August 16^{th} period of 2021 were 198% higher than between July 28^{th} and August 18^{th} of 2020. Rates in the August 16^{th} to September 7^{th} period of 2021 were 128% higher than between August 19^{th} to September 8^{th} , 2020.

The only significant correlation in 2020 between nitrate supply rate and N₂O fluxes occurred between the burial period of August 19th to September 8th and the N₂O fluxes from September 4th. No nitrate supply rates significantly correlated to N₂O fluxes in 2021.



Figure 3-13 Effect of land management practice – CT (contour tillage), DTGW (diversion terraces and grassed waterways), and DTGW+TD (diversion terraces, grassed waterways, and tile drainage) on nitrate supply rates measured in (top) 2020 and (bottom) 2021. The four sampling periods in 2020 were: (1) June 17 – July 12; (2) July 2 - July 28; (3) July 28 - August 18; and (4) August 19 – September 8. The three sampling periods in 2021 were: (1) July 26 - August 16; (2) August 16 – September 7; and (3) September 7 – September 22. Error bars represent the standard error of the mean (n = 12).

3.4 Discussion

3.4.1 Carbon Dioxide

The general emissions trend between years was different than that of N_2O . While N_2O emissions were minimal in 2020 and higher in 2019 and 2021, CO_2 emissions were, on average, higher in the hotter and drier 2020 than in 2021. Rising temperatures increase metabolic rates, leading to increased microbial respiration (Larcher, 2003). Temperature rises can increase CO_2 emissions exponentially up to a certain point, before decreasing due to enzyme denaturation or thermodynamic limitations (Fang & Moncrieff, 2001; Schipper et al., 2014). Temperature likely

partially explains why emissions were higher in 2020 than 2021. However, potatoes are associated with lower agricultural emissions relative to other major crops (Haile-Mariam et al., 2008; Jennings et al., 2020) and could also help explain the lowered emissions in 2021.

The frequent soil disturbances caused by tillage events in potato production cause a breakdown of soil aggregates, which aerates the soil and exposes organic matter to microbial decomposition (Grandy et al., 2002). In this study, we saw a decline in aggregate stability over the course of the two seasons. However, there were no statistically significant differences among LMPs and therefore likely does not contribute to explaining why CO₂ emissions in DTGW were significantly highest. Total carbon was higher in DTGW than DTGW+TD, but on par with CT. So, while this could contribute, it is likely not the only explanation. Soil temperature was similar in each of the terraced LMPs in each year (Figure 3-6), but CT had slightly lower soil temperatures in 2020. Soil VWC in 2020 was higher in DTGW+TD than in DTGW, but the results were reversed in 2021 (Figure 3-7). There is no obvious explanation, since the VWC levels reversing in each year is confounding, but the results are likely a combination of soil temperature, soil water, and total carbon.

3.4.2 Nitrous Oxide

The effects of the hot and dry summer of 2020 were evident: soil temperatures were high and soil moisture was low. This resulted in very low N₂O emissions. Conversely, in the 2019 and 2021 seasons, where soil water content was higher, N₂O emissions were higher. The N₂O emissions in this study are in line with emissions from conventional New Brunswick potato farming experiments that took place at the FRDC and ranged between 0 and 0.04 kg N₂O-N ha⁻¹ day⁻¹ (Burton et al., 2008).

Emissions by LMP did not follow the soil moisture trend. While DTGW had both the highest effluxes and soil moisture, CT had the lowest soil moisture, and DTGW+TD had the lowest average efflux – slightly below the conventional contour tillage and ~26% below DTGW.

The tile drainage appeared to help regulate soil water relative to its terraced counterpart (Figure 3-2). In both 2019 and 2021, soil moisture in DTGW+TD trended below the other LMPs. In 2021, specifically, VWC was nearly flat with no variability for nearly two months, while soil moisture consistently rose in the other LMPs. This water regulation from tile drainage likely aided in lowering N₂O effluxes. Denitrification is the dominant source of N₂O in humid climates such as Atlantic Canada (Mosier, 1998). When soil VWC approaches saturation, denitrification processes start to release more N₂O to the atmosphere under anaerobic conditions (Ball et al., 2008; Ruser et al., 2006). This commonly takes place at a water-filled pore space (WFPS) of 0.7 m³ m⁻³ (Bateman & Baggs, 2005) – or roughly a VWC of 0.36, assuming a bulk density of 1.3 g·cm³. VWC in DTGW+TD never eclipsed this value, while on 4 occasions VWC in DTGW approached or passed 0.4. The hypothesis was that we would see increased water content in both terraced LMPs, leading to higher N₂O effluxes. But while both LMPs had higher average VWC than CT, DTGW+TD emitted the least amount of N₂O.

Burton et al. (2012) stated that lower landscape positions have greater water contents for greater periods of time, leading to potentially greater denitrification and N₂O emissions. This was in contrast with results from this study – overall emissions were highest in upper slopes and lowest in lower slopes. This may be due to the fact that the hydrology was more complicated than initially thought (S. Li, personal communication, June 6, 2022). There may be water feed into the upper slope areas from above since the location of the site was not completely at the top of the knoll.

3.4.3 Methane

While there was slightly higher variability in 2019, CH₄ fluxes remained mostly negative through all seasons, regardless of soil moisture, resulting in all LMPs acting as C sinks in the growing period. Eastern Canadian soils are typically weak CH₄ sinks (Gregorich et al., 2005).

While the mean VWC values in DTGW and DTGW+TD only differed by ~3%, mean CH₄-C oxidation in DTGW+TD was roughly triple that of DTGW. This was, much like with N₂O, perhaps due to the variation seen in the VWC values in the wetter 2019 and 2021; DTGW+TD values were less variable than in DTGW and also had lower maximum values (Figure 3-2). Since CH₄-C uptake and soil moisture were positively correlated, it could be that the tile drainage – keeping the VWC more constant – was aiding in increasing the CH₄-C oxidation rates. Oxidation rates in the drier 2020 were almost identical in each LMP.

3.4.4 Nitrogen Supply Rate

Land management practice and slope position only had an effect on nitrate supply rates in 2020 and not 2021. There were two main differences between the growing seasons. First, 2020 was much hotter and drier than 2021 (Figure 3-2; 3-3). The crops were also different, with oats seeded in 2020 and potato seeded in 2021. Research has shown that soil ion fluxes measured with PRS-probes often decline with declining soil moisture (Qian & Schoenau, 2002). This aligns with the higher fluxes seen in the wetter 2021, but higher nitrogen (N) fertilization rates likely contributed as well. The N fertilization rate in 2021 after potato planting was 233% higher than after 2020 oat planting. This does not explain why there was no management effect in 2021.

It could be that since probes were placed in potato hills in 2021 that there was no effect seen. The potato rows might be more homogenous across LMPs since they are raised up above the furrow and the normal ground height. The differences in soil water between LMPs may not be as pronounced within the rows as they are 15 cm belowground where VWC was measured. The one curious result was in 2021 when, in the final burial period, the supply rate jumped considerably in CT but less so in the terraced LMPs. This does not align with the soil moisture trend, where DTGW was wettest and rates there should have risen the most if the rise was caused by an increase in soil moisture from rainfall.

Rates were initially high in 2020 and continued for six weeks, very likely due to the fertilizing that took place two days prior to the first installation of probes. The hot and dry summer was likely the source of the large drop in supply rate after July 28th, 2020. Supply rates from the first burial period in 2021 were nearly as high as from the first period of 2020, despite beginning 49 days after potato planting and fertilization and 39 days later relative to the first burial period in 2020. The N fertilization rates are high in potato farming, and the 190.5 kg N ha⁻¹ used on June 7th likely contributed to sustained nitrate supply through June and July. Higher supply rates were seen in 2021 in burial periods corresponding to similar dates in 2020.

3.5 Conclusions

DTGW+TD did not emit significantly more CO_2 in either year than CT – which served as the control LMP – despite soil temperature and moisture being slightly higher in 2020. DTGW+TD also emitted the least amount of N₂O on average and consumed the most CH₄ through the 3 summer seasons. In 2020, CT and DTGW+TD N₂O emissions were similar. But in 2021, where DTGW+TD had visibly more regulated VWC, emissions were even lower than in CT. DTGW emitted significantly more CO₂ in each year, despite all three LMPs having similar soil temperature and moisture in 2021.

The findings from this study show that soil moisture in DTGW+TD was more regulated than in CT or DTGW. In 2020, when the summer was hot and dry, tile drained plots surprisingly

had some of the highest VWC values. In 2021, when the field received more regular rainfall, DTGW+TD had the least amount of variability, and VWC did not trend upwards through the summer as much as in the other two LMPs. In 2019, 2020, and 2021, VWC in DTGW+TD was capped at roughly 0.3 m³ m⁻³, while the other two LMPs had values approaching 0.45 m³ m⁻³. Soil temperatures in 2020 were minimally cooler in CT, while in 2021 they were similar across all LMPs.

The hypothesis was that because of increased soil moisture from rainwater being directed downward into the soil and creating wetter soil environments, all GHG emissions would be elevated in the terraced LMPs. This was not the case, as DTGW+TD emitted less N₂O and CO₂, and consumed more CH₄, relative to DTGW. DTGW+TD also emitted less N₂O and consumed more CH₄ than CT. Carbon dioxide levels were on par between CT and DTGW+TD.

Chapter 4: Combined Effects of Nitrogen, Temperature, and Soil Water Content on GHG Emissions in a Short-Term Incubation Experiment

4.1 Introduction

The fundamental idea behind the LISWC system is to integrate several BMPs together such that they complement each other; the drawbacks of one BMP are offset by the benefits of another (Li, 2021). By implementing diversion terraces and grassed waterways, it essentially forces the usage of tillage conducted along the field contour. Up-and-down-slope oriented tillage becomes inefficient when the field segments are considerably wider than they are from top to bottom. The combination of diversion terraces, grassed waterways, and contour tillage reduces water erosion, but it can also lead to excessive amounts of water in the soil. This is then offset by the addition of tile drainage, which quickly removes excess water.

The pattern of greenhouse gas (GHG) emissions was anticipated to change based on the altered water regimes (and subsequent soil temperature changes) in the three land management practices (i.e., contour tillage; diversion terraces and grassed waterways; and diversion terraces, grassed waterways, and tile drainage). Soil water content and temperature are the two most influential parameters affecting soil gas emissions (Oertel et al., 2016; Raich & Schlesinger, 1992; Vicca et al., 2014). To study the effects of altered water regimes on greenhouse gas fluxes, the system must be experiencing regular rainfall. The summer of 2019, winter of 2019/2020, and summer of 2020 were all abnormally dry. The thought at the end of the 2020 growing season was that it might be difficult to detect differences with the *in-situ* field conditions; hence, a decision was made to complement the field seasons with an incubation experiment, in case the 2021 season mimicked the same hot and dry conditions. The objective of this study was to quantify the

GHG fluxes in response to volumetric water content (20%, 27%, 35%, 40% VWC), nitrogen fertilizer application rates (0, 170 kg N ha⁻¹), and temperature (10°C, 25°C) with a 6-day soil incubation experiment.

4.2 Materials and Methods

4.2.1 Experimental Design

In October 2020, composite bulk samples of a sandy loam texture were collected from a study site located at the Fredericton Research and Development Centre, with an average 55.3% sand, 34.5% silt, and 10.2% clay (Appendix B). The soils were taken using a Dutch auger (0-15 cm depth) from a northeast facing slope that faced the Saint John River (45°55'33.5"N 66°36'55.5"W). The soil was sieved through an 8-mm sieve to keep larger aggregates intact and remove large, organic debris or coarse fragments and was stored at 4°C until incubation was ready.

The incubation experiment was set up as a randomized complete block design. There was a total of 8 experimental units. Volumetric water content (VWC) had 4 factor levels: 20%, 27%, 35%, and 40%; equivalent to 15.5%, 20.8%, 27%, and 30.8% gravimetric water content (GWC) at an average bulk density of 1.3 g cm⁻³. Nitrogen fertilizer had 2 levels: 0 and 170 kg N ha⁻¹. The VWC levels were chosen to represent the range of soil moisture levels expected in an average growing season in central New Brunswick, as well as to examine higher moisture levels (40% VWC) that are less common. The N levels were chosen to represent a common fertilizer application rate used for potatoes in the region (Government of New Brunswick, n.d. -d). The two temperatures, 10°C and 25°C, were chosen to represent shoulder season temperatures and growing season temperatures, respectively.

The experiment was replicated 9 times for each temperature; all replicates for each separate temperature were run at the same time. Therefore, there were a total of 8 experimental units per replicate, and a total of 72 experimental units per temperature.

Microcosm preparation

The experiment used a method that assesses heterotrophic respiration using soil microcosm cores (Comeau et al., 2018). Each microcosm and filter paper, which covered the drainage holes, was weighed separately. Each microcosm was packed with field-moist soil to a bulk density of 1.3 g·cm⁻³, based on the respective gravimetric water contents. The microcosms were packed in two stages – half the soil mass was packed up to the 2.5 cm mark, followed by the remaining half, up to the 5 cm mark. This ensured optimized bulk density uniformity.

Each microcosm required either removal or addition of water to reach the desired VWC level. The microcosms which required drying were dried in the oven (max 30°C) until they reached the desired mass. All microcosms were dried a further 0.5 g to account for the future addition of either 0.5 mL water or 0.5 mL of N solution. Dry filter masses were recorded for all experimental units, whereas a wet filter mass correction was made uniformly across all 72 final microcosm masses based on an average of 5 wet filters. This was to balance out the water which will leave the soil and enter the dry filter.

Pre-incubation

To avoid a rewetting burst of greenhouse gases, soil moisture and temperature were established in advance. The samples were pre-incubated in a growth chamber (Conviron CMP 6050) at their respective temperatures for 3 days to allow soils to equilibrate. The microcosms were covered in parafilm for this time to prevent water evaporation. At the end of the pre-

incubation, if any microcosm masses were significantly lowered, a readjustment of water content levels occurred, and any sprouted seedlings were removed.

Nitrogen addition

Nitrogen solution was added to half the microcosms after the pre-incubation was performed. A nitrate-based 20-20-20 NPK fertilizer (TerraLink Horticulture Inc., Abbotsford, BC, Canada) was dissolved in distilled water to a concentration of 38.56 mg N·mL⁻¹. A total of 0.5 mL solution was added, which delivered the 19.28 mg N required. Each microcosm, regardless of treatment, was then adjusted individually to reach target soil moisture level. Microcosms were placed individually in hermetically sealed 2.9 L plastic containers. The containers were randomly located within the growth chamber.

4.2.2 Sampling and analysis

Soil GHG flux was measured 4 times over the course of the incubation, at time 0, 2, 4, and 6 days. Before the time 0 measurement, the 0.5 mL solutions of distilled water or fertilizer solution were added to respective treatments. Containers were flushed prior to the time 0 measurement to remove initial bursts of gases. 40 mL of ambient air was added immediately after flushing to balance out removing 20 mL of gas during each of the 4 samplings. When sampling, headspace mixing took place for 15 s to generate air flow and prevent stratification. After all samples were completed on a given day, their locations were randomized within the growth chamber again. All gas samples were analyzed at the University of British Columbia using a 7890A gas chromatograph (Agilent Technologies Inc., CA, USA) equipped with a flame ionization detector and electron capture detector and a PAL auto-sampler (Agilent Technologies Inc., CA, USA). The Ideal Gas Law was used to determine fluxes (Lang et al. 2011). The

greenhouse warming potentials (GWPs) of 298 for N_2O and 25 for CH_4 on a 100-year time scale were used to convert emissions to CO_2 equivalents (IPCC, 2019).

4.2.3 Statistical Analysis

The differences in GHG emissions and CO₂ equivalents between treatments were analyzed using ANOVA and Tukey post-hoc tests. A linear mixed-effects model was used and included VWC and N fertilization rate as fixed effects. Block was included as a random effect. Type III ANOVA was used to assess significance of the model interaction terms. A Tukey posthoc test using the *emmeans* function (Lenth, 2022) tested for differences in VWC and N fertilization rate. Analyses were conducted for each temperature separately. The normality and variance homogeneity assumptions of ANOVA were assessed using Shapiro-Wilks test and Bartlett's test, respectively, in addition to diagnostic plots. A statistical significance threshold of $\alpha = 0.05$ was used for hypothesis testing. Statistical analyses were conducted using R software version 4.0.3 (R Core Team 2020). Transformations were made on N₂O fluxes to meet model assumptions.

4.3 Results

Carbon dioxide

Only N fertilization rate had significant effects on CO₂ fluxes at each temperature (Figure 4-1). There was no interaction between VWC and N fertilizer rate at either temperature. No significant differences were observed between VWC levels at 10°C. At 25°C, there were significant differences between: 20% and 27%; 20% and 35%; 20% and 40%; and 27% and 40% VWC (Table 4-1). Nitrogen fertilizer levels significantly differed at each temperature (p < 0.001). At 25°C, the increase in VWC linearly increased CO₂ fluxes up to 34%, after which it plateaued.

Nitrous oxide

For N₂O, there were significant interactions between VWC and N fertilization rate at both temperatures (p < 0.001). There were significant (p < 0.001) differences in N₂O fluxes between fertilized and non-fertilized treatments in 40% VWC treatments at 10°C and 27%, 35%, and 40% VWC treatments at 25°C (Figure 4-2). All VWC treatments below 40% with 0 N fertilizer had slightly negative fluxes, regardless of temperature. The only 170 kg N ha⁻¹ treatment that had negative fluxes was at 20% VWC and 10°C. The combination of 40% VWC and fertilizer addition resulted in the largest positive fluxes at each temperature.

Methane

For CH₄, at 10°C, there was moderate evidence of a significant interaction between VWC and N fertilization rate (p = 0.037). However, no treatments significantly differed (Figure 4-3). At 25°C, there was no significant VWC × N fertilizer rate interaction, VWC was not significant, and N fertilizer rate significance was suggestive but not conclusive (p = 0.08). All CH₄ mean fluxes were negative, regardless of temperature or treatment.

CO₂ equivalents

When comparing CO₂ equivalents of all three gases, there was no significant interaction (p > 0.4) between VWC and N fertilization rate at either temperature (Figure 4-4). Neither fertilization rate (p > 0.12) nor VWC (p > 0.3) were significant at either temperature. Though not statistically significant, the CO₂ equivalents emitted at 25°C and 40% VWC were 199% higher than the next highest emitting treatment (25°C and 35%).

		10°C			25°C	
Contrast	df (degrees of freedom)	t	p-value	df	t	p-value
20-27	56	0.42	0.975	56	-2.62	0.053
20-35	56	-1.85	0.264	56	-4.18	< 0.001
20-40	56	-1.79	0.289	56	-5.59	<.0001
27-35	56	-1.92	0.233	56	-1.97	0.213
27-40	56	-1.88	0.250	56	-2.68	0.046
35-40	56	0.01	1.000	56	0.02	1.000
20-35 20-40 27-35 27-40 35-40	56 56 56 56 56	-1.85 -1.79 -1.92 -1.88 0.01	0.264 0.289 0.233 0.250 1.000	56 56 56 56 56	-4.18 -5.59 -1.97 -2.68 0.02	<0.00 <.000 0.21 0.04 1.00

Table 4-1 Pairwise comparisons of volumetric water content (VWC) groups, averaged over N fertilization level, using Tukey adjustment.



Figure 4-1 Mean CO_2 flux calculated at the end of the 6-day incubation experiment. Error bars represent the standard error of the mean (n = 9).



Figure 4-2 Mean N₂O flux calculated at the end of the 6-day incubation experiment. Error bars represent the standard error of the mean (n = 9). For each volumetric water content (VWC) within a temperature, asterisks denote a significant difference between nitrogen fertilizer rate × VWC treatments ($\alpha = 0.05$).



Figure 4-3 Mean CH_4 flux calculated at the end of the 6-day incubation experiment. Error bars represent the standard error of the mean (n = 9).



Figure 4-4 Total CO₂ equivalent emissions over the 6-day incubation experiment. Error bars represent the standard error of the mean (n = 9).

4.4 Discussion

Carbon dioxide

Volumetric water content did not have an effect on CO_2 fluxes at 10°C, where temperature clearly had more of an effect on soil respiration than soil water content. It appears that soil respiration was not sensitive to changes in VWC at this temperature. This suggests that in shoulder seasons there may be little change to soil CO_2 fluxes in between 20 and 40% VWC levels. However, CO_2 flux could drop near saturation, where VWC would approach 50.7% (100% WFPS), or below 20%, where respiration would be water-limited.

Water content did have a significant effect on fluxes at 25°C; as VWC increased, CO_2 fluxes rose steadily up to 35% and peaked. It is possible fluxes would drop at a higher VWC than 40%. A study of an Ontario soil with similar soil particle distribution found fluxes to be maximal at 65% WFPS (Fairbairn, 2020) – equivalent to 33% VWC in our soils. In my study, fluxes did

not decline significantly between 35 and 40%. It can be very near saturation that some soils finally begin to show a decline in CO₂ fluxes. An incubation study on a fine loamy soil in Germany found that CO₂ emissions only became restricted past 98% WFPS (Ruser et al., 2006). Our soil was a well-aggregated sandy loam with 55% sand. In the field, this soil would have difficulty reaching VWC values greater than 40% in summer temperatures, especially on sloping land, as observed in 2021.

The addition of N fertilizer had a significant effect on CO₂ fluxes at both temperatures. Nitrogen addition raised CO₂ fluxes by approximately 422% at 10°C and 55% at 25°C, on average, across all treatments. Nitrogen addition raised CO₂ fluxes by an average 892% between temperatures. Increases in soil N content will generally lead to increases in soil respiration, as long as C is not a limiting factor (Niu et al., 2010; Peng et al., 2011). This N addition can lead to higher soil respiration sensitivity toward soil moisture and to lowered sensitivity toward temperature (Peng et al., 2011). This holds true at 25°C; however, at 10°C there was no sensitivity toward VWC increasing.

Nitrous Oxide

Three quarters of the N₂O fluxes at 0 kg N ha⁻¹ were slightly negative. Only 40% VWC treatments were positive, where fluxes increased from $7.6 \pm 9.6 \,\mu g \, N_2 O$ -N kg⁻¹ day⁻¹ without N addition to $30.7 \pm 11.2 \,\mu g \, N_2 O$ -N kg⁻¹ day⁻¹ with N addition. Negative N₂O fluxes are uncommon and often research centers on net N₂O production (Chapuis-Lardy et al., 2007). However, there is a wide range of evidence of net negative fluxes measured in various conditions from temperate to tropical areas and from natural to agricultural systems (Chapuis-Lardy et al., 2007). For 20, 27, and 35% VWC levels at 0 kg N ha⁻¹ and both temperatures, the slightly negative fluxes were likely a result of the 6-day incubation length. The N₂O concentration inside

the container headspace dropped in most treatment combinations from day 0 to day 2, before steadily increasing upward. For these six specific treatments, concentrations were approaching the starting concentration at day 6, and, given a slightly longer incubation, would have flipped positive. The fertilized treatment at 10°C and 20% VWC, however, consumed substantially more N₂O in the 6 days. The same concentration pattern occurred in this treatment, but it would have taken considerably longer than 6 days to flip positive. The initial reduction in N₂O concentration between day 0 and 2 could have been caused by a number of factors. Consumption of N₂O is often due to low mineral N availability or large moisture contents, but there are other factors such as nitrifier denitrification, aerobic denitrification, soil pH, and O₂ content (Chapuis-Lardy et al., 2007).

The greatest N₂O emissions were observed in the treatment with highest temperature, water content, and N fertilization rate. Fluxes reached $1065 \pm 269 \,\mu g \, N_2 O$ -N kg⁻¹ day⁻¹. The dominant N₂O production process in humid climates is denitrification (Mosier, 1998), so it was not surprising to see highest emissions in near-saturated soil conditions of both temperatures.

Methane

All treatments led to uptake of methane, and no CH₄ emissions were observed. Methane in soils is produced by methanotrophic microorganisms under strictly anaerobic conditions (Dutaur & Verchot, 2007; Topp & Pattey, 1997). Even at the maximum 40% VWC studied, no CH₄ was emitted. I had hypothesized that we would see emissions beginning at 40%. At 25°C there was suggestive evidence of lowered uptake of CH₄ in the presence of N fertilizer. Addition of inorganic N fertilizers has the ability to reduce CH₄ uptake or increase CH₄ emission, and this could explain the potential increase in fertilized treatments (Bodelier & Laanbroek, 2004).

CO₂ equivalents

Carbon dioxide was primarily responsible for total GHG emissions in CO₂ equivalent units when total emissions were positive in a treatment over the 6 day period. Approximately 75% of the total equivalent in the treatment consisting of 170 kg N ha⁻¹ and 40% VWC at 25°C was due to N₂O emissions. The one negative occurrence of total CO₂ equivalents occurred in the 20% VWC and 0 kg N ha⁻¹ treatment at 10°C; the negative totals from N₂O and CH₄ slightly offset the positive CO₂ emissions. Overall, CO₂ equivalents generally rose with increasing soil water content.

4.5 Conclusions

The 6-day soil incubation under different VWC, N fertilizer levels, and temperatures had varying effects on GHG emissions. Only at 25°C did VWC have a significant effect on CO_2 fluxes, while N addition had a significant effect at both temperatures. The addition of N had a significant effect on N₂O fluxes. At the lowest VWC and temperature, there was a net uptake in N₂O over the 6 days. At each temperature, the treatments of 40% VWC and N addition resulted in the highest fluxes. There was suggestive evidence that N addition reduced CH₄ uptake at 25°C. Otherwise, there were no significant treatment effects on CH₄ fluxes.

There is interest in the results that correspond to periods where field measurements were not able to be collected – that is, in the early spring and later in the fall where soil is likely to be wetter due to reduced evaporation. This is potentially a critical period for N₂O and CH₄, when saturated soil, and consequently anoxic conditions, can produce higher emissions. In the 10°C incubation, only N₂O showed higher sensitivity toward water content. There is no evidence that at cooler soil temperatures, CO₂ or CH₄ emissions would rise significantly. Since incubation studies are uncommon at 10°C, results from this incubation study provide insights into how

temperature, soil water content, and nitrogen fertilizer affect GHG emissions from typical agricultural soil in Atlantic Canada.

Chapter 5: General Conclusions and Recommendations for Future Research

To address soil erosion and soil health concerns, as well as to address the uncertainties of temperature and precipitation pattern changes in the wake of a changing climate, the LISWC system has been proposed in the last five years as a new paradigm for soil and water conservation in agricultural landscapes. The LISWC system combines diversion terraces, grassed waterways, tile drainage, water retention structures, supplemental irrigation, conservation tillage practices, and soil-landscape restoration, with an overarching goal of integrating these BMPs to obtain all the benefits while minimizing the drawbacks. The objective of my thesis study was to evaluate the effects of terraces, grassed waterways, and tile drainage relative to contour tillage on various soil quality indicators and greenhouse gas (GHG) emissions during the initial years of management implementation. I also conducted a short-term incubation experiment to quantify GHG fluxes in response to volumetric water content, nitrogen fertilizer application rates, and temperature. Diversion terraces should have an effect on soil water content, thereby affecting GHG emissions, but a continuous range of soil water content is difficult to produce in the field and therefore justifies an incubation study.

5.1 Soil Aggregate Stability and Labile Carbon Response to Second and Third Years of Contour Tillage, Diversion Terrace, Grassed Waterway, and Tile Drainage

Implementation

There was no statistically significant change in total soil organic carbon between the second and third years since management implementation. No change was expected, as total carbon typically changes slowly. This justifies the use of indicators that are more sensitive to changes in land management that may show early changes in the soil ecosystem.

Aggregate stability declined between post-planting 2020 and post-harvest 2021 in all three LMPs, but there was no significant difference between the three LMPs. However, DTGW+TD saw much more variability than the other two, and this was the only LMP showing any evidence of increasing aggregate stability. The increases in labile C in both DTGW and DTGW+TD should improve the physical structure of the soil and improve aggregation. It is possible that there hasn't been enough time since the initial construction of the study site to see any improvements in aggregate stability.

Land management practice had a significant effect on POXC. The contour tillage saw a decrease in POXC over the study period, dropping 11.2% from post-planting 2020 to post-harvest 2021, while DTGW increased an average of 19.8% and DTGW+TD increased an average of 50.6%. POXC is considered to be a more processed fraction of labile C, closely correlating with SOC. It is possible that over time DTGW and DTGW+TD will begin to see a significant increase in SOC storage compared to before the BMPs were implemented.

The density fractionation of particulate organic matter led to interesting results, especially in the mineral-associated organic matter (MAOM) fraction. The contour tillage lost an average of $0.6 \pm 1.4 \text{ g C kg}^{-1}$ soil (-3.75%) in the MAOM fraction, while DTGW+TD gained an average 1.7 $\pm 0.9 \text{ g C kg}^{-1}$ soil (+9.7%). The largest change was seen in DTGW, where C in the MAOM fraction rose by an average of $2.5 \pm 1.0 \text{ g C kg}^{-1}$ soil, or about +15.8%. Almost 80% of the C in this soil was found in this fraction. The rise in MAOM C in both terraced LMPs was not likely due to any breakdown of aggregates since the pattern of aggregate stability decrease was mostly similar in all LMPs. The breakdown of fresh organic material (fPOM) may be being immediately captured within this fraction; the pattern was similar between LMPs, but the terraced LMPs may have been more efficient at capturing these low molecular weight compounds. The MAOM fraction represents a more stable form of soil C due to its association with soil mineral particles and protection from decomposition. It is possible that the LISWC system is able to increase stabilized C, thereby increasing the health of the soil, since that fraction constitutes the largest amount of C and also had the largest C increases.

As expected, the fPOM fraction, consisting of the freshest, least decomposed material, declined in all LMPs between post-planting 2020 and post-harvest 2021. CT lost the most C from this fraction. This is likely due to small, free light particles washing out of the field in large rainfalls and during the spring snow thaw because of the lack of soil berms. The tile drained LMP, DTGW+TD, lost the least amount of C in the fPOM fraction. This may be due to the fact that quicker removal of subsurface water kept the levels of microbial decomposition lower relative to the other LMPs.

The greatest amount of C lost from the oPOM fraction was also observed in CT. This fraction of most New Brunswick soils has the oldest C (unpublished data) and is therefore important to protect from decreasing (L.P. Comeau, personal communication, March 30, 2022). There was a small gain of 0.22 ± 0.1 g C kg soil⁻¹ in DTGW+TD, perhaps due to a small increase in microaggregates that remained on top of the 53 µm sieve; the occluded particulate organic matter is controlled mostly by physical protection from decomposition in aggregates. The only aggregate size fraction I analyzed was 1-2 mm, and there may have been an increase in smaller-sized micro-aggregates.

Overall, both drained and undrained terraced LMPs offered more promising results for the improvement of soil health, relative to contour tillage, in the second and third years of implementation. Both DTGW and DTGW+TD saw increases in the labile C fraction POXC – a fraction that is significantly related to SOC. Both also saw large increases in MAOM – a fraction

that has shown positive feedback between new SOM stabilization from labile C and total soil C. These results occurred despite all three LMPs undergoing a decrease in aggregate stability. An increase in SOM would be positive for degraded New Brunswick soils and would improve soil functioning by improving soil physical structure and increasing biological activity.

5.2 Greenhouse Gas Emissions and Nitrate Supply Rate Response to Initial Three Years of Diversion Terrace, Grassed Waterway, and Tile Drainage Implementation

In this study, average CO₂ fluxes were highest in the undrained DTGW in both 2020 and 2021, though not statistically different than CT in 2020. However, cumulative CO₂ emissions were greater in DTGW than in either of the other LMPs, true for both years. CT and DTGW+TD did not significantly differ in either year in terms of CO₂ fluxes. In 2020, CO₂ fluxes were 63% higher than in 2021. This was likely caused by a much hotter summer of 2020 that increased microbial metabolism and stimulated the decomposition of organic matter.

Land management practice had a significant effect on N₂O fluxes. In 2019, DTGW+TD fluxes were significantly lower than either CT or DTGW. Fluxes were higher in DTGW in 2020 and 2021, though not statistically significant. This likely corresponds to the higher VWC in DTGW, decreasing the oxygen levels within the soil and increasing denitrification rates. Tile drainage kept soil moisture levels lower than in the other LMPs. This was particularly clear in 2021, where VWC stayed relatively constant while VWC increased each time we measured gas fluxes using the collars.

There were no significant differences between LMPs for CH_4 fluxes or cumulative CO_2 equivalents of CH_4 in 2019, 2020, or 2021. Aside from DTGW in 2019, which was skewed by one large spike, all CH_4 fluxes were negative in any LMP each year. It is common in the area for soils to act as methane sinks.

This is the first study in the area measuring GHG emissions from erosion control structures that include diversion terraces, grassed waterways, and tile drainage. Soil organic carbon in agricultural ecosystems in New Brunswick has been declining for decades. This study demonstrated that the addition of tile drainage to diversion terraces and grassed waterways significantly reduced the loss of stored carbon (in the form of CO_2) relative to undrained diversion terraces and grassed waterways, while also not emitting significantly more carbon than contour tillage, in the three years after implementation. The results were similar with respect to the loss of nitrogen in the form of harmful N₂O, where undrained diversion terraces and grassed waterways generally emitted more N₂O in the first three years after implementation. This has implications for farmers with respect to the efficiency of added inputs and may allow for the reduction of nitrogen fertilizer.

5.3 Combined Effects of Nitrogen, Temperature, and Soil Water Content on GHG Emissions in a Short-Term Incubation

To supplement the field experiment, a 6-day soil incubation experiment was conducted to investigate how volumetric water content, temperature, and nitrogen fertilization rate interact to affect greenhouse gas emissions (GHG) in a sandy loam soil from the FRDC research station. Nitrogen fertilization rate and VWC had a significant effect on CO₂ fluxes at 25°C, where they rose until 35% VWC, then plateaued. However, at 10°C, only N fertilization had an effect on CO₂ fluxes, and VWC did not alter CO₂ fluxes whatsoever. This suggests that in colder periods in spring or fall, soil moisture is not the dominant factor that controls CO₂. A field consisting of diversion terraces that happen to increase soil moisture would most likely not see increased CO₂ fluxes relative to a field without erosion control structures.

There were significant effects on N₂O fluxes. At 10°C, the only significant difference occurred at 40% VWC, where fluxes increased between fertilized and non-fertilized soils. At 25°C, fluxes increased significantly between fertilized and non-fertilized soils at 27%, 35%, and 40%. The combination of 40% VWC and fertilizer addition resulted in a large spike in emissions above the unfertilized 40% VWC treatment, rising to 1065 ± 269 from $30.7 \pm 11.1 \,\mu g \, N_2 O - N \, kg^{-1} \, day^{-1}$ – an increase of 3369%. Approximately 75% of the total CO₂e in the treatment consisting of 170 kg N ha⁻¹ and 40% VWC at 25°C was due to N₂O emissions. This has implications for the timing of fertilization in the field, as emissions can be greatly reduced if avoiding this combination of conditions. To minimize the risk of a large burst of N₂O, fertilizer application should not occur when soil VWC is at or above 40%.

5.4 Recommendations for Future Research

The results of this study do not consider the spring snow melt period, which is often the wettest period of the year in the study region. It would be useful to examine periods during which soils approach saturation, to determine whether these terraced systems affect spring GHG emissions. Emissions can be significant outside the growing season in eastern Canada, often exceeding in-season emissions (Ruser et al., 2006; Wagner-Riddle & Thurtell, 1998). No studies have examined GHG emission response to snow melt periods under diversion terrace and grassed waterway management systems.

Another possibility would be to assess the GHG emissions coming from the furrow in the seasons in which potato is grown. I placed collars directly into the potato ridges, but water is not uniformly distributed in the potato ridge-furrow complex (Burton et al., 2012). Water application from rainfall will tend to accumulate in the furrow, leading to higher water-filled pore space and decreased O₂ diffusion into the soil. Combined with a higher bulk density, this leads to higher

denitrification and N_2O emissions. The results in this study could be underestimating the overall emissions by not accounting for furrows.

The CH₄ oxidation rates in CT and DTGW+TD were similar to those seen in potato ridges in a German experiment (Ruser et al., 1998). Another study (Flessa et al., 2002) found ~86% of the CH₄-C uptake occurred in the potato ridges. This suggests most of the uptake was likely captured in our study but, again, that furrows are unaccounted for. The same study found that uncompacted tractor rows were a C sink but that compacted tractor rows were a source of CH₄, so, like N₂O, future research in this site should attempt to study differences between ridges and compacted/uncompacted furrows to avoid over- or underestimation of fluxes.

There is evidence that tile drainage can increase macroaggregation but negatively affect other aggregate size classes (Kumar et al., 2014). As we only studied aggregates in the 1 to 2 mm size class, it might be interesting to investigate smaller size classes and their response to a tiledrained LISWC system. This might help to explain why there was no significant difference in our study between the effect of LMPs on aggregate stability, but the distribution of carbon was changing (i.e., MAOM increasing in terraced LMPs).

Research has also shown that tile drainage can lead to high levels of nitrate loss in potato fields (Madramootoo et al., 1992). Indirect N_2O emissions from the water retention pond could add to the fluxes emitted from soil and push emissions higher. This could negate the observed benefits from the LISWC system on N_2O emissions relative to a contour-tilled field.

Bibliography

- Abid, M., & Lal, R. (2009). Tillage and drainage impact on soil quality: II. tensile strength of aggregates, moisture retention and water infiltration. *Soil & Tillage Research*, 103(2), 364-372. https://doi.org/10.1016/j.still.2008.11.004
- Adler, R. L., Singh, G., Nelson, K. A., Weirich, J., Motavalli, P. P., & Miles, R. J. (2020). Cover crop impact on crop production and nutrient loss in a no-till terrace topography. *Journal* of Soil and Water Conservation, 75(2), 153-165. https://doi.org/10.2489/jswc.75.2.153
- Agriculture and Agri-Food Canada. (2012). Diversion terraces and grassed waterways in hilly potato land: Studies explore benefits and reveal possible trade-offs of these popular practices. Agriculture and Agri-Food Canada.
- Agriculture and Agri-Food Canada. (2020). Potato Market Information Review 2019-2020. Retrieved from https://agriculture.canada.ca/en/canadas-agriculturesectors/horticulture/horticulture-sector-reports/potato-market-information-review-2019-2020#a1.2
- Angers, D. A., Edwards, L. M., Sanderson, J. B., & Bissonnette, N. (1999). Soil organic matter quality and aggregate stability under eight potato cropping sequences in a fine sandy loam of Prince Edward Island. *Canadian Journal of Soil Science*, 79(3), 411-417. https://doi.org/10.4141/S98-033
- Bakker, M.M., Govers, G., & Rounsevell, M.D.A., (2004). The crop productivity–erosion relationship: an analysis based on experimental work. Catena 57, 55–76.
- Bakker, M. M., Govers, G., Kosmas, C., Vanacker, V., Oost, K. v., & Rounsevell, M. (2005). Soil erosion as a driver of land-use change. *Agriculture, Ecosystems & Environment,* 105(3), 467-481. https://doi.org/10.1016/j.agee.2004.07.009

- Ball, B. C., Crichton, I., & Horgan, G. W. (2008). Dynamics of upward and downward N.sub.2O and CO.sub.2 fluxes in ploughed or no-tilled soils in relation to water-filled pore space, compaction and crop presence. *Soil & Tillage Research*, 101(1-2), 20. https://doi.org/10.1016/j.still.2008.05.012
- Baryla, A., & Pierzgalski, E. (2008). Ridged terraces—functions, construction and use. Journal of Environmental Engineering and Landscape Management 16 (2), la–lf.
- Bateman, E. J., & Baggs, E. M. (2005). Contributions of nitrification and denitrification to N2O emissions from soils at different water-filled pore space. *Biology and Fertility of Soils*, 41(6), 379-388. https://doi.org/10.1007/s00374-005-0858-3
- Bodelier, P. L. E., & Laanbroek, H. J. (2004). Nitrogen as a regulatory factor of methane oxidation in soils and sediments. *FEMS Microbiology Ecology*, 47(3), 265-277. https://doi.org/10.1016/S0168-6496(03)00304-0
- Bongiorno, G., Bünemann, E. K., Oguejiofor, C. U., Meier, J., Gort, G., Comans, R., Mäder, P., Brussaard, L., & de Goede, R. (2021). Corrigendum to "Sensitivity of labile carbon fractions to tillage and organic matter management and their potential as comprehensive soil quality indicators across pedoclimatic conditions in Europe" [ecol. indic. 99 (2019) 38–50]. *Ecological Indicators, 121*, 107093 https://doi.org/10.1016/j.ecolind.2020.107093
- Burton, D., Zebarth, B., Gillam, K., & MacLeod, J. (2008). Effect of split application of fertilizer nitrogen on N₂O emissions from potatoes. *Canadian Journal of Soil Science*, 88(2), 229-229.
- Burton, D. L., Zebarth, B. J., McLeod, J. A., & Goyer, C. (2012). Nitrous oxide emissions from potato production and strategies to reduce them. *Sustainable potato production: Global*

case studies (pp. 251-271). Springer Netherlands. https://doi.org/10.1007/978-94-007-4104-1_14

- Caron, J. Kay, B. D., & Stone, J. A. (1992). Improvement of structural stability of a clay loam with drying. *Soil Science Society of America Journal*, 56(5), 1583-1590. https://doi.org/10.2136/sssaj1992.03615995005600050041x
- Carter, M. R. (2002). Soil quality for sustainable land management: Organic matter and aggregation interactions that maintain soil functions. *Agronomy Journal*, 94(1), 38-47. https://doi.org/10.2134/agronj2002.3800
- Carter, M. R., Gregorich, E. G., Angers, D. A., Donald, R. G., & Bolinder, M. A. (1998).
 Organic C and N storage, and organic C fractions, in adjacent cultivated and forested soils of eastern Canada. *Soil & Tillage Research*, 47(3), 253-261.
 https://doi.org/10.1016/S0167-1987(98)00114-7
- Carter, M. R., & Sanderson, J. B. (2001). Influence of conservation tillage and rotation length on potato productivity, tuber disease and soil quality parameters on a fine sandy loam in eastern Canada. *Soil & Tillage Research*, 63(1), 1-13. https://doi.org/10.1016/S0167-1987(01)00224-0
- Carter, M.R. (2002). Soil quality for sustainable land management. Agron. J. 94:38–47. doi:10.2134/agronj2002.0038
- Carter, M. R., Noronha, C., Peters, R. D., & Kimpinski, J. (2009). Influence of conservation tillage and crop rotation on the resilience of an intensive long-term potato cropping system: Restoration of soil biological properties after the potato phase. *Agriculture, Ecosystems & Environment, 133*(1), 32-39. https://doi.org/10.1016/j.agee.2009.04.017

- Chapuis-Lardy, L., Wrage, N., Metay, A., Chotte, J., & Bernoux, M. (2007). Soils, a sink for N2O? A review. *Global Change Biology*, 13(1), 1-17. https://doi.org/10.1111/j.1365-2486.2006.01280.x
- Chow, T.L., J.L. Daigle, I. Ghanem, & H. Cormier. (1990). Effects of potato cropping practices on water runoff and soil erosion. *Canadian Journal of Soil Science* 70:137-148.
- Chow, T. L., Rees, H. W., & Daigle, J. L. (1999). Effectiveness of terraces/grassed waterway systems for soil and water conservation: A field evaluation. *Journal of Soil and Water Conservation*, 54(3), 577-583.
- Chow, T.L., H.W. Rees, & J. Monteith. (2000). Seasonal distribution of runoff and soil loss under four tillage treatments in the upper St. John River valley New Brunswick, Canada. *Canadian Journal of Soil Science* 80:649-660.
- Chow, L., Rees, H., Xing, Z. (2010). Impacts of long-term intensive potato production and conservation terraces/grassed waterway on runoff hydrology and soil quality. In Proceedings of the 19th World Congress of Soil Science: Soil Solutions for a Changing World, Brisbane, Australia, 1–6 August 2010; pp. 68–71.
- Comeau, L., Lai, D. Y. F., Cui, J. J., & Hartill, J. (2018). Soil heterotrophic respiration assessment using minimally disturbed soil microcosm cores. *MethodsX*, 5, 834-840. https://doi.org/10.1016/j.mex.2018.07.014
- Coote, D. R., Dumanski, J. & Ramsey, J. F. (1981). An assessment of the degradation of agricultural lands in Canada. Land Resource Research Institute, Research Branch, Agriculture Canada, Ottawa, ON. 86 pp.
- Culman, S. W., Snapp, S. S., Freeman, M. A., Schipanski, M. E., Beniston, J., Lal, R., Drinkwater, L. E., Franzluebbers, A. J., Glover, J. D., Grandy, A. S., Lee, J., Six, J.,

Maul, J. E., Mirksy, S. B., Spargo, J. T., & Wander, M. M. (2012). Permanganate Oxidizable Carbon Reflects a Processed Soil Fraction that is Sensitive to Management. *Soil Science Society of America Journal*, 76(2), 494–504. doi:10.2136/sssaj2011.0286

- Dalal, R. C., & Mayer, R. J. (1986a). Long term trends in fertility of soils under continuous cultivation and cereal cropping in southern Queensland. III distribution and kinetics of soil organic carbon in particle-size fractions. *Aust. J. Soil Res.* 24, 293–300.
- Dalal, R. C., & Mayer R J (1986b). Long term trends in fertility of soils under continuous cultivation and cereal cropping in southern Queensland. IV Loss of organic carbon from different density functions. *Aust. J. Soil Res.* 24, 301–309.
- Delisle, J. (2018). *Effect of biochar on soil quality and potato productivity in New Brunswick, Canada.* (Master's thesis). Retrieved from https://www.library.ubc.ca/
- Dutaur, L., & Verchot, L. V. (2007). A global inventory of the soil CH4sink: A global inventory of the soil CH4 sink. *Global Biogeochemical Cycles*, 21(4), n/a. https://doi.org/10.1029/2006GB002734
- Edwards, L., Richter, G., Bernsdorf, B., Schmidt, R., & Burney, J. (1998). Measurement of rill erosion by snowmelt on potato fields under rotation in Prince Edward Island (Canada). *Canadian Journal of Soil Science*, 78(3), 449-458. https://doi.org/10.4141/S97-053
- Eijkelkamp. (2008). Method 08.13 wet sieving apparatus operating instructions. Eijkelkamp Soil and Water, the Netherlands. https://en.eijkelknamp.com/products/laboratoryequipment/wet-sieving-apparatus.html (accessed 20 July 2021)
- Eilers, W., L. Mackay, L. Graham, & A. Lefebvre, editor. (2010). Environmental sustainability of Canadian agriculture. Rep. No. 3. Canada Dep. of Agriculture. http://publications.gc.ca/site/archivee-archived.
html?url=http://publications.gc.ca/collections/collection_2011/agr/A22-201-2010-eng.pdf (accessed 20 Apr. 2022)

Elliott, E. T. (1986). Aggregate structure and carbon, nitrogen, and phosphorus in native and cultivated soils. *Soil Science Society of America Journal*, *50*(3), 627-633. https://doi.org/10.2136/sssaj1986.03615995005000030017x

Fang, C., & Moncrieff, J. B. (2001). The dependence of soil CO2 efflux on temperature. Soil Biology & Biochemistry, 33(2), 155-165. https://doi.org/10.1016/S0038-0717(00)00125-5

- Fairbairn, L. G. (2020). *Linking soil moisture content and carbon dioxide fluxes: From batch experiments to process-based modelling*. Retrieved from https://www.library.ubc.ca/
- Flessa, H., Ruser, R., Schilling, R., Loftfield, N., Munch, J. C., Kaiser, E. A., & Beese, F. (2002). N2O and CH4 fluxes in potato fields: Automated measurement, management effects and temporal variation. *Geoderma*, 105(3-4), 307-325. https://doi.org/10.1016/S0016-7061(01)00110-0
- Fraser, H., & Fleming, R. (2001). Environmental benefits of tile drainage: Literature review. Retrieved from http://rainalgoma.ca/wp-content/uploads/2014/09/fleming_drainage.pdf
- Fuss, R., Hueppi, R., & Pedersen, A.R. (2020). 'gasfluxes: Greenhouse Gas Flux Calculation from Chamber Measurements'. R package. Available at: https://cran.rproject.org/web/packages/gasfluxes/index.html
- Gagnon, B., Lalande, R., & Fahmy, S. H. (2001). Organic matter and aggregation in a degraded potato soil as affected by raw and composted pulp residue. *Biology and Fertility of Soils*, 34(6), 441-447. https://doi.org/10.1007/s00374-001-0428-2
- Govaere, L., Morin, M. D., Frigault, J. J., Boquel, S., Cohen, A., Lamarre, S. G., & Morin, P. J. (2019). Transcriptome and proteome analyses to investigate the molecular underpinnings

of cold response in the colorado potato beetle, leptinotarsa decemlineata. *Cryobiology*, 88, 54-63. https://doi.org/10.1016/j.cryobiol.2019.03.011

Government of Canada. (2019). Canadian Climate Normals 1981-2010 Station Data. Retrieved from https://climate.weather.gc.ca/

Government of Canada. (2021). Historical Data. Retrieved from

https://climate.weather.gc.ca/historical_data/search_historic_data_e.html

- Government of New Brunswick. (n.d. -a). Soil Management. Retrieved from https://www2.gnb.ca/content/gnb/en/departments/10/agriculture/content/crops/potatoes/s oil_management.html
- Government of New Brunswick. (n.d. -b). Soil Erosion. Retrieved from https://www2.gnb.ca/content/gnb/en/departments/10/agriculture/content/land_and_enviro nment/environmental_sustainability/soil_erosion.html
- Government of New Brunswick. (n.d. -c). Beneficial management practices for reducing runoff and soil loss. Retrieved from: https://agriculture.canada.ca/en/agriculture-andenvironment/agriculture-and-water/watershed-protection/watershed-evaluationbeneficial-management-practices/beneficial-management-practices-reducing-runoff-andsoil-loss

Government of New Brunswick. (n.d. -d). *Nitrogen Management for Potato: General Fertilizer Recommendations*. Retrieved from https://www2.gnb.ca/content/dam/gnb/Departments/10/pdf/Agriculture/potato_general_fa ctsheet.pdf

Govers, G., Quine, T. A., Desmet, P. J. J., & Walling, D. E. (1996). The relative contribution of soil tillage and overland flow erosion to soil redistribution on agricultural land. *Earth*

Surface Processes and Landforms, 21(10), 929-946. https://doi.org/10.1002/(SICI)1096-9837(199610)21:10<929::AID-ESP631>3.0.CO;2-C

- Govers, G., Vandaele, K., Desmet, P., Poesen, J., & Bunte, K. (1994). The role of tillage on soil redistribution on hillslopes. *European Journal of Soil Science*, 45(4), 469-478. https://doi.org/10.1111/j.1365-2389.1994.tb00532.x
- Grandy, A. S., Porter, G. A., & Erich, M. S. (2002). Organic amendment and rotation crop effects on the recovery of soil organic matter and aggregation in potato cropping systems. *Soil Science Society of America Journal*, 66(4), 1311-1319. https://doi.org/10.2136/sssaj2002.1311
- Gregorich, E. G., Carter, M. R., Angers, D. A., Monreal, C. M., & Ellert, B. H. (1994). Towards a minimum data set to assess soil organic matter quality in agricultural soils. *Canadian Journal of Soil Science*, 74(4), 367-385. https://doi.org/10.4141/cjss94-051
- Gregorich, E. G., Rochette, P., Vandenbygaart, A. J., & Angers, D. A. (2005). Greenhouse gas contributions of agricultural soils and potential mitigation practices in eastern Canada:
 Greenhouse gas contributions and mitigation potential in agricultural regions of North America. *Soil & Tillage Research*, *83*(1), 53-72.
- Gregorich, E. G., Carter, M. R., Angers, D. A., & Drury, C. F. (2009). Using a sequential density and particle-size fractionation to evaluate carbon and nitrogen storage in the profile of tilled and no-till soils in eastern Canada. *Canadian Journal of Soil Science*, 89(3), 255-267. https://doi.org/10.4141/CJSS08034
- Haddix, M. L., Gregorich, E. G., Helgason, B. L., Janzen, H., Ellert, B. H., & Francesca Cotrufo,M. (2020). Climate, carbon content, and soil texture control the independent formation

and persistence of particulate and mineral-associated organic matter in soil. *Geoderma*, *363*, 114160. https://doi.org/10.1016/j.geoderma.2019.114160

- Haile-Mariam, S., Collins, H., & Higgins, S. (2008). Greenhouse gas fluxes from an irrigated sweet corn (*Zea mays* L.)-potato (*Solanum tuberosum* L.) rotation. *J. Environ. Qual.* 37, 759–771. doi: 10.2134/jeq2007.0400
- Haynes, R. J. (2005). Labile organic matter fractions as central components of the quality of agricultural soils: An overview. *Advances in agronomy* (pp. 221-268). Elsevier Science & Technology. https://doi.org/10.1016/S0065-2113(04)85005-3
- Haynes, R. J., & Tregurtha, R. (1999). Effects of increasing periods under intensive arable vegetable production on biological, chemical and physical indices of soil quality. *Biology* and Fertility of Soils, 28(3), 259-266. https://doi.org/10.1007/s003740050491
- Hinds, R. P. (1989). Soil physical requirements for potato production in Atlantic Canada.Atlantic Advisory Committee on Soil Resource Management, Atlantic ProvincesAgricultural Services Co-ordinating Committee. 9 pp.
- Hoegh-Guldberg, O., Jacob, D., Taylor, M., Guillén Bolaños, T., Bindi, M., Brown, S.,
 Camilloni, I. A., Diedhiou, A., Djalante, R., Ebi, K., Engelbrecht, F., Guiot, J., Hijioka,
 Y., Mehrotra, S., Hope, C. W., Payne, A. J., Pörtner, H., Seneviratne, S. I., Thomas, A., .
 ... Zhou, G. (2019). The human imperative of stabilizing global climate change at 1.5°C.
 Science (*American Association for the Advancement of Science*), 365(6459), 1263-+.
 https://doi.org/10.1126/science.aaw6974
- Huffman, R.L., D.D. Fangmeier, W.J. Elliot, & S.R. Workman. (2013). Soil and Water Conservation Engineering, 7th Edition. *American Society of Agricultural and Biological Engineers*. ISBN 1-892769-86-7.

- Hurisso, T. T., Culman, S. W., Horwath, W. R., Wade, J., Cass, D., Beniston, J. W., Bowles, T. M., Grandy, A. S., Franzluebbers, A. J., Schipanski, M. E., Lucas, S. T., & Ugarte, C. M. (2016). Comparison of Permanganate-Oxidizable Carbon and Mineralizable Carbon for Assessment of Organic Matter Stabilization and Mineralization. *Soil Science Society of America Journal*, 80(5), 1352–1364. doi:10.2136/sssaj2016.04.0106
- IPCC. (2019). Climate Change and Land: an IPCC special report on climate change, desertification, land degradation, sustainable land management, food security, and greenhouse gas fluxes in terrestrial ecosystems [P.R. Shukla, J. Skea, E. Calvo Buendia, V. Masson-Delmotte, H.-O. Pörtner, D. C. Roberts, P. Zhai, R. Slade, S. Connors, R. van Diemen, M. Ferrat, E. Haughey, S. Luz, S. Neogi, M. Pathak, J. Petzold, J. Portugal Pereira, P. Vyas, E. Huntley, K. Kissick, M. Belkacemi, J. Malley, (eds.)]. In press.
- Jacinthe, P. A., Lal, R., & Kimble, J. M. (2001). Organic carbon storage and dynamics in croplands and terrestrial deposits as influenced by subsurface tile drainage. *Soil Science*, *166*(5), 322-335. https://doi.org/10.1097/00010694-200105000-00003
- Jennings, S. A., Koehler, A. K., Nicklin, K. J., Deva, C., Sait, S. M., & Challinor, A. J. (2020). Global potato yields increase under climate change with adaptation and CO2 fertilisation. *Frontiers in Sustainable Food Systems*, 4, 519324.
- Kumar, S., Nakajima, T., Mbonimpa, E. G., Gautam, S., Somireddy, U. R., Kadono, A., Lal, R., Chintala, R., Rafique, R., & Fausey, N. (2014). Long-term tillage and drainage influences on soil organic carbon dynamics, aggregate stability and corn yield. *Soil Science and Plant Nutrition (Tokyo), 60*(1), 108-118. https://doi.org/10.1080/00380768.2013.878643

- Lal, R., Mahboubi, A. A., & Fausey, N. R. (1994). Long-Term tillage and rotation effects on properties of a central Ohio soil. *Soil Science Society of America Journal*, 58(2), 517-522. https://doi.org/10.2136/sssaj1994.03615995005800020038x
- Lang, M., Cai, Z., & Chang, S. X. (2011). Effects of land use type and incubation temperature on greenhouse gas emissions from Chinese and Canadian soils. *Journal of Soils and Sediments*, 11(1), 15-24. https://doi.org/10.1007/s11368-010-0260-0
- Langmaid, K. K., MacMillan, J. K., & Losier, J. G. (1980). Soils of Madawaska County. New Brunswick. Res. Branch, Canada Dep. of Agric. and New Brunswick Dep. of Agric., Fredericton, NB.
- Langmaid, K. K., MacMillan, J. K. & Losier, J. G. (1976). Soils of northern Victoria County, New Brunswick. Seventh Report of the New Brunswick Soil Survey. Canada Department of Agriculture, Cat. No. A57_176/1976. New Brunswick Department of Agriculture and Rural Development, Fredericton, NB. 152 pp.
- Larcher, W. (2003). Physiological plant ecology: Ecophysiology and stress physiology of functional groups (4th ed.). Springer.
- Lavallee, J. M., Soong, J. L., & Cotrufo, M. F. (2020). Conceptualizing soil organic matter into particulate and mineral-associated forms to address global change in the 21st century. *Global Change Biology*, 26(1), 261-273. https://doi.org/10.1111/gcb.14859
- Le Bissonnais, Y. (1988). Comportement d'agrégats terreux soumis à l'action de l'eau: analyse des mécanismes de désagrégation. Agronomie (Paris) 8: 915–924.
- Lenth, R. (2022). 'emmeans': Estimated marginal means, aka least-squares means. Retrieved from https://cran.r-project.org/package=emmeans.

- Li, S. (2021). Landscape integrated soil and water conservation (LISWC) system for sloping landscapes in Atlantic Canada. *Agriculture (Basel)*, 11(5), 427. https://doi.org/10.3390/agriculture11050427
- Lindstrom, M. J., Nelson, W. W., Schumacher, T. E., & Lemme, G. D. (1990). Soil movement by tillage as affected by slope. *Soil & Tillage Research*, 17(3), 255-264. https://doi.org/10.1016/0167-1987(90)90040-K
- Lobb, D. A., Kachanoski, R. G., & Miller, M. H. (1995). Tillage translocation and tillage erosion on shoulder slope landscape positions measured using 137Cs as a tracer. *Canadian Journal of Soil Science*, 75(2), 211-218. https://doi.org/10.4141/cjss95-029
- Madramootoo, C., Wiyo, K., & Enright, P. (1992). Nutrient losses through tile drains from two potato fields. *Applied Engineering in Agriculture*, 8(5), 639-646. https://doi.org/10.13031/2013.26136
- Madramootoo, C. A., Johnston, W. R., Ayars, J. E., Evans, R. O., & Fausey, N. R. (2007).
 Agricultural drainage management, quality and disposal issues in North America.
 Irrigation and Drainage, 56(S1), S35-S45. https://doi.org/10.1002/ird.343
- Milburn, P., & Gartley, C. (1988). Subsurface drainage and land use in New Brunswick. Can. Agric. Eng, 30, 13-17. Mosier, A. R. (1998). Soil processes and global change. Biology and Fertility of Soils, 27(3), 221-229. https://doi.org/10.1007/s003740050424
- Milburn, P. Richards, J. E., Gartley, C., Pollock, T., O'Neill, H., & Bailey, H. (1990). Nitrate leaching from systematically tiled potato fields in new brunswick, canada. *Journal of Environmental Quality*, 19(3), 448-454.

https://doi.org/10.2134/jeq1990.00472425001900030016x

- Niu, S., Wu, M., Han, Y.I., Xia, J., Zhang, Z.H., Yang, H., & Wan, S., (2010). Nitrogen effects on net ecosystem carbon exchange in a temperate steppe. *Glob. Change Biol*.16, 144– 155.
- Ochuodho, T. O., Olale, E., Lantz, V. A., Damboise, J., Chow, T. L., Meng, F., ... & Li, S.
 (2013). Impacts of soil and water conservation practices on potato yield in northwestern New Brunswick, Canada. *Journal of Soil and Water Conservation*, 68(5), 392-400.
- Oertel, C., Matschullat, J., Zurba, K., Zimmermann, F., & Erasmi, S. (2016). Greenhouse gas emissions from soils—A review. *Chemie Der Erde*, 76(3), 327-352. https://doi.org/10.1016/j.chemer.2016.04.002
- Paustian, K. et al. (2004). Agricultural mitigation of greenhouse gases: science and policy options. Council on Agricultural Science and Technology (CAST) report, R141 2004, ISBN 1-887383-26-3, p. 120, May, 2004.
- Peng, Q., Dong, Y., Qi, Y., Xiao, S., He, Y., Ma, T. (2011). Effects of nitrogen fertilization on soil respiration in temperate grassland in Inner Mongolia, China. *Environ. Earth Sci.* 62, 1163–1171.
- Poeplau, C., Don, A., Six, J., Kaiser, M., Benbi, D., Chenu, C., Cotrufo, M. F., Derrien, D.,
 Gioacchini, P., Grand, S., Gregorich, E., Griepentrog, M., Gunina, A., Haddix, M.,
 Kuzyakov, Y., Kühnel, A., Macdonald, L. M., Soong, J., Trigalet, S., . . . Lawrence
 Berkeley National Lab. (LBNL), Berkeley, CA (United States). (2018). Isolating organic
 carbon fractions with varying turnover rates in temperate agricultural soils A
 comprehensive method comparison. *Soil Biology & Biochemistry*, *125*(C), 10-26.
 https://doi.org/10.1016/j.soilbio.2018.06.025

- PP Systems. (2018). EGM-5 Portable CO₂ Gas Analyzer Operation Manual. Retrieved from http://ppsystems.com/download/technical_manuals/80109-1-EGM-5_Operation_V103.pdf
- Qian, P., & Schoenau, J. J. (2002). Practical applications of ion exchange resins in agricultural and environmental soil research. *Canadian Journal of Soil Science*, 82(1), 9-21. https://doi.org/10.4141/S00-091
- Quine, T. A., Govers, G., Walling, D. E., Xinbao, Z., Desmet, P. J. J., Yusheng, Z., & Vandaele,
 K. (1997). Erosion processes and landform evolution on agricultural land; new
 perspectives from caesium-137 measurements and topographic-based erosion modelling. *Earth Surface Processes and Landforms*, 22(9), 799-816.
- R Core Team (2020). R: A language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria. URL https://www.R-project.org/
- Raich, J. W., & Schlesinger, W. H. (1992). The global carbon dioxide flux in soil respiration and its relationship to vegetation and climate. *Tellus. Series B, Chemical and Physical Meteorology*, 44(2), 81-99. https://doi.org/10.3402/tellusb.v44i2.15428
- Rasiah, V., Kay, B. D., & Martin, T. (1992). Variation of structural stability with water content: Influence of selected soil properties. *Soil Science Society of America Journal*, 56(5), 1604-1609. https://doi.org/10.2136/sssaj1992.03615995005600050044x
- Rees, H.W., & Fahmy, S.H. (1984). Soils of the Agriculture Canada Research Station Fredericton, N.B. LRRI Research Branch Agriculture Canada. Retrieved from https://sis.agr.gc.ca/cansis/publications/surveys/nb/nb994/index.html
- Rees, H. W., Chow, T. L., & Gregorich, E. G. (2007). Spatial and temporal trends in soil properties and crop yield at a site under intensive up- and down-slope potato production

in northwestern New Brunswick. *Canadian Journal of Soil Science*, 87(4), 383-398. https://doi.org/10.4141/CJSS07017

- Rees, H. W., Chow, T. L., & Gregorich, E. G. (2008). Soil and crop responses to long-term potato production at a benchmark site in northwestern New Brunswick. *Canadian Journal of Soil Science*, 88(3), 409-422. https://doi.org/10.4141/CJSS07086
- Rees, H. W., Chow, T. L., Walker, D. F., & Smith, A. O. M. (1999). Potential use of underseeded barley to increase carbon inputs to a loam soil in the New Brunswick potato belt. *Canadian Journal of Soil Science*, 79(1), 211-216. https://doi.org/10.4141/S98-012
- Rees, H., Chow, L., Zebarth, B., Xing, Z., Toner, P., Lavoie, J., & Jaigle, J. L. (2014). Impact of supplemental poultry manure application on potato yield and soil properties on a loam soil in north-western New Brunswick. *Canadian Journal of Soil Science*, 94(1), 131017051323003-65. https://doi.org/10.4141/CJSS2013-009
- Ritter, W. F., Rudra, R. P., Milburn, P. H., & Prasher, S. (1995). Drainage and water quality in northern United States and eastern Canada. *Journal of irrigation and drainage engineering*, 121(4), 296-301.
- Rolfe, C. J. (1993). Using Subsidies to Promote Environmental Protection in Agriculture. Vancouver: West Coast Environmental Law Research Foundation.
- Ruser, R., Flessa, H., Schilling, R., Steindl, H., & Beese, F. (1998). Soil compaction and fertilization effects on nitrous oxide and methane fluxes in potato fields. *Soil Science Society of America Journal*, 62(6), 1587-1595.

https://doi.org/10.2136/sssaj1998.03615995006200060016x

Ruser, R., Flessa, H., Russow, R., Schmidt, G., Buegger, F., & Munch, J. C. (2006). Emission of N2O, N2 and CO2 from soil fertilized with nitrate: Effect of compaction, soil moisture

and rewetting. *Soil Biology & Biochemistry*, *38*(2), 263-274. https://doi.org/10.1016/j.soilbio.2005.05.005

- Saini, G.R., Grant, W.J. (1980). Long-term effects of intensive cultivation on soil quality in the potato-growing areas of New Brunswick (Canada) and Maine (U.S.A.). *Can. J. Soil Sci.* 60, 421–428.
- Schipper, L. A., J. K. Hobbs, S. Rutledge, & V. L. Arcus. (2014). Thermodynamic theory explains the temperature optima of soil microbial processes and high Q10 values at low temperatures. *Global Change Biology*, 20(11), 3578–3586.
- Sierra, C. A., Malghani, S., & Loescher, H. W. (2017). Interactions among temperature, moisture, and oxygen concentrations in controlling decomposition rates in a boreal forest soil. *Biogeosciences*, 14(3), 703-710. https://doi.org/10.5194/bg-14-703-2017
- Sikora, L. J., Yakovchenko, V., Cambardella, C. A., & Doran, J. W. (1996). Assessing soil quality by testing organic matter. In F. R. Magdoff, M. A. Tabatabai & E. A. Hanlon (Eds.), *Soil organic matter: Analysis and interpretation* (pp. 41-50). Soil Science Society of America. https://doi.org/10.2136/sssaspecpub46.c5
- Six, J., Elliott, E. T., Paustian, K., & Doran, J. W. (1998). Aggregation and soil organic matter accumulation in cultivated and native grassland soils. *Soil Science Society of America Journal*, 62(5), 1367-1377. https://doi.org/10.2136/sssaj1998.03615995006200050032x

Smith, P., Martino, D., Cai, Z., Gwary, D., Janzen, H., Kumar, P., McCarl, B., Ogle, S., O'Mara, F., Rice, C., Scholes, B., Sirotenko, O., Howden, M., McAllister, T., Pan, G., Romanenkov, V., Schneider, U., Towprayoon, S., Wattenbach, M., & Smith, J. (2008).
Greenhouse gas mitigation in agriculture. *Philosophical Transactions. Biological Sciences, 363*(1492), 789-813. https://doi.org/10.1098/rstb.2007.2184

- Stark, J.C., & Porter, G.A. (2005). Potato nutrient management in sustainable cropping systems. Am. J. Potato Res. 82: 329–338. doi: https://doi.org/10.1007/BF02871963
- Statistics Canada. (2021). Table 32-10-0358-01 Area, Production and Farm Value of Potatoes. Retrieved from: https://doi.org/10.25318/3210035801-eng
- Stocker, T.F., D. Qin, G.-K. Plattner, L.V. Alexander, S.K. Allen, N.L. Bindoff, . . . S.-P. Xie, (2013). Technical Summary. In: Climate Change 2013: The Physical Science Basis.
 Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Stocker, T.F., D. Qin, G.-K. Plattner, M. Tignor, S.K. Allen, J. Boschung, A. Nauels, Y. Xia, V. Bex and P.M. Midgley (eds.)].
 Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.
- Stuart, V. (2017). Watershed Evaluation of Beneficial Management Practices (WEBs):
 Managing our Land and Protecting our Water Through Long-Term Watershed-Scale
 Research: Final Report (2004-2013). Agriculture and Agri-Food Canada, Ottawa. Ont.
- Tiessen, K. H. D., Li, S., Lobb, D. A., Mehuys, G. R., Rees, H. W., & Chow, T. L. (2009). Using repeated measurements of 137Cs and modelling to identify spatial patterns of tillage and water erosion within potato production in Atlantic Canada. *Geoderma*, 153(1), 104-118. https://doi.org/10.1016/j.geoderma.2009.07.013
- Topp, E., & Pattey, E. (1997). Soils as sources and sinks for atmospheric methane. *Canadian Journal of Soil Science*, 77(2), 167-177. https://doi.org/10.4141/S96-107
- Unger, PW. (1992). Infiltration of simulated rainfall: tillage system and crop residue effects. *Soil Science Society of America Journal* 56:283-289.
- Van Oost, K., Van Muysen, W., Govers, G., Deckers, J. & Quine, T. A. (2005). From water to tillage erosion dominated landform elements. *Geomorphology* 72:193-203.

- Vicca, S., Bahn, M., Estiarte, M., van Loon, E. E., Vargas, R., Alberti, G., Ambus, P., Arlain, M. A., Beier, C., Bentley, L. P., Borken, W., Buchmann, N., Collins, S. L., De Dato, G., Dukes, J. S., Escolar, C., Fay, P., Guidolotti, G., Hanson, P. J., . . . Janssens, I. A. (2014). Can current moisture responses predict soil CO2 efflux under altered precipitation regimes? A synthesis of manipulation experiments. *Biogeosciences, 11*(11), 2991-3013. https://doi.org/10.5194/bg-11-2991-2014
- von Lützow, M., Kögel-Knabner, I., Ekschmitt, K., Flessa, H., Guggenberger, G., Matzner, E., & Marschner, B. (2007). SOM fractionation methods: Relevance to functional pools and to stabilization mechanisms. *Soil Biology & Biochemistry*, *39*(9), 2183-2207. https://doi.org/10.1016/j.soilbio.2007.03.007
- Wagner-Riddle, C. & Thurtell, G. W. (1998). Nitrous oxide emissions from agricultural fields during winter and spring thaw as affected by management practices. *Nutrient Cycling in Agroecosystems*, 52(2-3), 151-163. https://doi.org/10.1023/A:1009788411566
- Wander, M. (2004). Soil organic matter fractions and their relevance to soil function. *Soil organic matter in sustainable agriculture*. CRC Press, Boca Raton, FL, 67-102.
- Wang, C., Rees, H. W. and Daigle, J.-L. (1984). Classification of Podzolic soils as affected by cultivation. *Canadian Journal of Soil Science*. 64: 229-239.
- Wardle, D. A. (1992). A comparative assessment of factors which influence microbial biomass carbon and nitrogen levels in soil. *Biological reviews*, 67(3), 321-358.
- Wilson, C., Zebarth, B. J., Burton, D. L., & Goyer, C. (2018). Short-Term effects of diverse compost products on soil quality in potato production. *Soil Science Society of America Journal*, 82(4), 889-900. https://doi.org/10.2136/sssaj2017.10.0345

- Weil, R. R., Stine, M. A., Islam, K. R., Gruver, J. B., & Samson-Liebig, S. E. (2003). Estimating active carbon for soil quality assessment: A simplified method for laboratory and field use Estimating active carbon for soil quality assessment: A simplified method for laboratory and field use. *American Journal of Alternative Agriculture*, *18*(1), 3–17. doi:10.1079/AJAA200228
- Wilson, C., Zebarth, B. J., Burton, D. L., & Goyer, C. (2018). Short-Term effects of diverse compost products on soil quality in potato production. *Soil Science Society of America Journal*, 82(4), 889-900. https://doi.org/10.2136/sssaj2017.10.0345
- Yang, Q., Meng, F., Zhao, Z., Chow, T. L., Benoy, G., Rees, H. W., & Bourque, C. P. (2009).
 Assessing the impacts of flow diversion terraces on stream water and sediment yields at a watershed level using SWAT model. *Agriculture, Ecosystems & Environment*, 132(1-2), 23-31. https://doi.org/10.1016/j.agee.2009.02.012
- Zebarth, B. J., Moreau, G., Dixon, T., Fillmore, S., Smith, A., Hann, S., & Comeau, L. (2022). Soil properties and topographic features influence within-field variation in potato tuber yield in New Brunswick, Canada. *Soil Science Society of America Journal*, 86(1), 134-145. https://doi.org/10.1002/saj2.20342
- Zelazny, V.E. (Ed). (2007). Our landscape heritage: the story of ecological land classification in New Brunswick. New Brunswick Department of Natural Resources. Fredericton, New Brunswick, Canada.
- Zielke, R. C., & Christenson, D. R. (1986). Organic carbon and nitrogen changes in soil under selected cropping systems. *Soil Science Society of America Journal*, 50(2), 363-367. https://doi.org/10.2136/sssaj1986.03615995005000020022x

Appendices

Appendix A Photos of the LISWC system at the FRDC

A.1 CT (left), DTGW (middle), DTGW+TD (right) with oats in the field in 2020.



A.2 View from the side of one of the grassed waterways with the soil berm on the downslope edge (left).



A.3 View of DTGW+TD plots (foreground) from the side.



A.4 View of CT from the bottom looking upward after a heavy rainfall in which visible signs of water erosion appear.



Appendix B List of soil properties for each LMP and slope position. Bulk density was

		Soil property						
		рН	CEC	Sand	Silt	Clay	Bulk density	с
			meq/100g	%	%	%	g.cm^3	%
LMP	Slope position							
СТ	Upper	5.43 (0.056)	10.3 (0.21)	52.2 (0.6)	37.7 (0.33)	10.3 (0.33)	1.26 (0.048)	1.83 (0.084)
СТ	Mid	5.45 (0.043)	10 (0.37)	57.3 (0.42)	33.2 (0.4)	9.33 (0.21)	1.23 (0.037)	1.78 (0.076)
СТ	Lower	5.52 (0.04)	10.2 (0.17)	57.2 (0.4)	32.2 (0.31)	10.8 (0.17)	1.35 (0.055)	2.01 (0.046)
DTGW	Upper	5.58 (0.07)	9.5 (0.22)	53.8 (0.79)	36.3 (0.67)	9.67 (0.49)	1.28 (0.055)	1.78 (0.078)
DTGW	Mid	5.48 (0.054)	10 (0.26)	55 (0.37)	34.2 (0.4)	10.8 (0.17)	1.4 (0.038)	1.75 (0.052)
DTGW	Lower	5.45 (0.056)	11 (0.37)	55.3 (0.61)	32.8 (0.7)	11.8 (0.17)	1.22 (0.032)	2.03 (0.031)
DTGW+TD	Upper	5.65 (0.072)	8.17 (0.31)	55.2 (0.48)	36 (0.68)	8.83 (0.48)	1.25 (0.057)	1.37 (0.063)
DTGW+TD	Mid	5.07 (0.042)	6.67 (0.33)	55.7 (0.67)	34.5 (0.5)	9.83 (0.31)	1.21 (0.044)	1.57 (0.063)
DTGW+TD	Lower	5.17 (0.042)	7.67 (0.42)	55.7 (0.76)	33.3 (0.49)	10.7 (0.21)	1.35 (0.096)	1.96 (0.092)

determined August 19th and 26th, 2020. All others determined July 2nd, 2020.



Appendix C Timeline of events in 2020



Appendix D Timeline of events in 2021

Disking took place on May 20th. Rocks were picked following spring disking by approximately a dozen people on June 3rd. A first harrowing took place on May 26th. A final harrowing took place on June 7th, followed by a Russet Burbank potato planting. Rows were spaced at 0.91 m, and plants were spaced at 0.38 m within rows. An application of 17-17-17 urea-based fertilizer was applied during planting at a rate of 1000 lbs acre⁻¹ (190.5 kg N ha⁻¹). This was slightly lower than the 208 kg N ha⁻¹ fertilizer rate that is the typical maximum fertilization rate for potato farming in New Brunswick (Government of New Brunswick, n.d. -d).

On June 14th, a pre-emergent herbicide (Sencor 1.5 kg/ha and Dual II 1.75 L/ha) was sprayed to suppress weed activity. Potato rows were hilled on July 20th. Insecticide was sprayed on both July 23rd and August 11th to suppress Colorado Potato Beetle (*Leptinotarsa decemlineata*) activity. Fungicide (Admire) spraying was scheduled every Friday (weather permitting) but sometimes took place a day earlier or later depending on rains. Topkill (Reglon) was applied on September 23rd. Top flail took place on October 4th, and potato harvest finally took place on October 6th. All plots were cleaned with a rock picker on October 7th. The final disking took place on October 19th.

Appendix E Rill erosion after Hurricane Elsa.



Appendix F Soil Aggregate Stability Analysis Procedure

The method separates the soil sample into loose aggregate, stable aggregate, and sand portions. A total of 4g of soil was used per sample, and samples were run in duplicate. Loose aggregate portion was determined by first placing the soil in the sieve nests of the apparatus, lowering them into the metal cans below that were filled with distilled water, and allowing to rest for 5 minutes. The sieves were then lowered into the starting position, and a 3-minute process begins, whereby the sieves are oscillated up and down gently to allow loose aggregate portion to collect in the metal can below. Water and soil were then vacuum filtered over pre-weighed 90mm filters in a Buchner funnel. Paper and soil were washed into a pre-weighed and labeled tin to dry in the oven at 110°C. At this point, stable aggregate and sand portions remain in the sieve. For the stable aggregate portion, the metal cans were filled with a dispersing solution (sodium hydroxide) and the sieves were oscillated in the apparatus for 5 minutes. After the 5 minutes, the sieve was raised, and a rubber-tipped policeman was used to gently break up aggregates. These oscillation and physical agitation steps were repeated twice more so that the aggregates were fully dispersed. A 55mm glass filter was used to vacuum filter the stable aggregate portion into pre-weighed and numbered tins to dry in the oven. Remaining sand portion in the sieve was washed into a final tin for oven-drying. If the difference between the final aggregate stability percentage values of reps 1 and 2 was greater than 7%, a third rep was required.



Appendix G Soil Aggregate Stability by LMP in 2020 and 2021.

Appendix H Permanganate Oxidizable Carbon Analysis Procedure

A 0.2 M stock solution of potassium permanganate (KMNO₄) was first prepared from a solution of KMNO₄ in solution of calcium chloride (CaCl₂). This was then used to create the 0.02 M working solution for the procedure. It should be noted that the creation of a working solution differs from the initial protocol developed by Weil (2003) and Culman (2013). This was to help minimize variation between samples caused by instrument or human error.

Four standard concentrations of KMnO₄ were prepared from the 0.2 *M* stock solution (0.005 M, 0.01 M, 0.015 M, and 0.02 M). These standards, and their resulting absorbance values (once diluted to final working standards), are then used to build a standardized curve. The intercept and slope values from this curve are then used in the final calculation of permanganate oxidizable carbon.

A total of 2.5 g of air-dried soil was sub-sampled from each of the 54 subplot samples and placed in a centrifuge tube. Working solution (20ml) was added to each tube and placed in an oscillating shaker for two minutes at 240 rpm. Each tube was then placed in the dark for 10 minutes to settle. Once settled, 0.5 ml of supernatant was pipetted into a glass test tube with 49.5 ml DDH₂O, capped, and inverted to mix thoroughly. A total of ~1.5 ml of the final sample solution was pipetted into disposable cuvettes for absorbance analysis in a Biochrom Libra spectrophotometer (Biochrom Ltd., Cambridge, United Kingdom). Samples were analyzed at 550 nm, and calculated using the following formula:

 $POXC (mg kg^{-1}) = [0.02 mol/L - (a+b*Abs)] * (9000 mg C/mol) * (0.02 L solution/Wt)$ where 0.02 mol/L is the initial solution concentration, a is the intercept of the standard curve, b is the slope of the standard curve, Abs is the Absorbance of Unknown, 9000 is the milligrams of carbon oxidized by 1 mole of MnO_4 changing from Mn^{7+} to Mn^{2+} , 0.02 L is the volume of stock solution reacted, and Wt is the weight of air-dried soil sample in kg.

Appendix I Density Fractionation Analysis Procedure

Density fractionation was performed on soil samples from: July 2nd, 2020; September 24th, 2020; May 27th, 2021; and October 7th, 2021. The protocol was based on the method described in Peoplau et al. (2018) with one alteration – removal of the sieving at 0.2 mm (Appendix G). The result of the protocol was three specific fractions: fPOM (free light POM separated from oPOM); oPOM (sand-sized and occluded POM, 50-200 μ m); and MAOM (silt-and clay-sized mineral associated organic matter, < 50 μ m rinsed from catch pan).

A total of 100 g of air-dried, 2mm-sieved soils were first shaken for 16 hours at 225 RPM on an orbital shaker in plastic bottles with 200 mL distilled water and seven 5mm glass beads. When shaking was finished, contents were wet-sieved at 50 µm. The MAOM fraction passed below to the catch pan after rinsing with distilled water. The remaining portion on top of the sieve were the combined fPOM and oPOM fractions. The MAOM fraction was rinsed into one beaker, while the remaining fractions were rinsed into a second beaker. All beakers were placed in the oven at 55°C overnight to fully dry.

A total of between 50 and 100mL of sodium iodide (specific gravity of 1.7 g/mL) is added to the fPOM and oPOM fractions. In this case, vigorous stirring for 1 minute was substituted for shaking. After a roughly 2-minute settling time, the floating material was fPOM and the sunken material was the mineral oPOM fraction. A spoon was used to scoop as much floating fPOM as possible onto a filter paper inside a Buchner funnel, before carefully emptying remaining fPOM and sodium iodide (NaI) onto the filter paper, without allowing any mineral fraction from the bottom of the beaker to transfer into the funnel. Pure sodium iodide was then collected with vacuum filtration for re-use. The fPOM was rinsed with distilled water and vacuum filtered to ensure no NaI remained in the fraction. The fPOM was then rinsed off the

116

filter paper into a 100mL beaker. The rinsing and vacuum filtering was repeated with the remaining mineral oPOM fraction. Both beakers with respective fractions were oven dried at 55°C overnight. Once dry, the MAOM and oPOM fractions were ground in a Retsch ball mill grinder for 3 minutes. The fPOM fraction was hand-ground using a mortar and pestle. Final samples were analyzed at the FRDC for total C concentrations using an Elementar varioMACRO apparatus (Elementar Americas Inc., Mt. Laurel, NJ).



Appendix J Diagram of modified protocol for density fractionation

Appendix K Collar set up for collecting GHG samples in 2019, 2020, and 2021



Appendix L COVID-19 notes from my workbook

Below are the notes I wrote down in my masters workbook beginning in early 2020. I was slated to fly to Ottawa at the end of April to pick up my parents' car for the summer and drive out to Fredericton, but of course the pandemic shutdown happened in March. Maja joked about keeping a journal, and I figured it could be an interesting thing to look back on decades from now. March and April were uncertain times, and many people were scrambling to figure out what to do with growing season arriving quickly.

These are random, unorganized thoughts and notes of mine, with no particular structure – only what I found notable at any given time.

NBA shutdown - 1st realization this was going to start affecting people outside Wuhan and Washington

University of Washington shutdown - still wasn't thinking too much about it affecting us. Chinese community was aware and malls in Richmond began to empty long before Canadian shutdown.

Classes moved online. Fortunately, no exam in FRST545. Exam cancelled in FRST530. Assignment grade only. Very lucky! Many others having to adapt to online for 6 long weeks. TA'ing online a bit more annoying. No more labs :(

The Great Toilet Paper Shortage of 2020 :/

Began dating online over 'Zoom'. Had two first dates on Zoom followed by bike ride second dates. Dated Johanna for about 3 weeks before leaving. Gave me a kickstand for my bike the day before I left. Bittersweet. Hope to see her in fall. Will see if I date anyone in Fredericton. What a time to be dating!

Counting myself very fortunate at this time. Research maybe delated. But still on way to New Brunswick where things are in very good shape. Only 118 cases and no deaths. Many friends having to make contingency plans. Raelani was thinking about quitting grad school :(but is still subletting my place. Amy Mays is basically having to throw away a year. People are having to apply for exemptions to conduct research at UBC. Rumours that September classes will be <u>only</u> online. Wow. Hard to fathom.

Think I will apply to the BCIA in the next couple weeks. Economy is in the tank. Thinking this could last potentially years, so it will coincide with graduation from UBC. Might need the leg up on folks. Have 2 weeks in Freddy to do my best quarantine.

30 people on the flight heading east!! There are more people on our Friday trivia zoom gatherings (40+ people) than on this flight.

The Drive to NB.

Everything was closed along the way so had to get inventive with where to go... #2. Added a little extra time driving around looking for forests to use.

Went through a checkpoint somewhere in Quebec. Somewhere between Quebec City and NB border. Pretty quick. Checkpoint at NB border was about 10 mins. They were pretty thorough. Took roommate's info, Louis's info, job info, etc.

Had a laugh passing a sign for a potato museum.

Immediately passes another sign "Town of Florenceville: French fry capital of the world" ... funny, must be in NB now.

5 days into quarantine. Had to run outside today. Feeling antsy.

Roommate brought home Covered Bridge chips and Chicken Bones – two NB delicacies.

After over 2 weeks of no new cases there were apparently 2 new ones today. Still, we are very lucky here. Saw a tent set up downtown yesterday with a few people dressed head to toe in PPE.

Struggling mightily with motivation for school. Did my taxes yesterday but schoolwork has been absent for weeks on end. Moving was one thing. Covid doesn't help. There is certainly time. But finding self-discipline right now is hard. Gone are the routines. The workplace environment. Etc.

May 20th. Had orientation meeting today over Zoom with new students. There is talk of a regular social meeting on Zoom which would go a long way to making things more exciting around here.

Was supposed to go out to the field for the 1st time today, but it got cancelled at the last minute

May 26th. A glimmer of hope today. The email from Josée Owen seems to suggest I'll be able to at least take EGM samples soon.

June 18th. Have been able to gas sample and plant probes. No sign of collars or soil sampling. But have gone for beers and met people which helps.

June 22nd. Louis said early July I might have access to the lab. Great news! Time to get busy!

July 9th. Lab access pushed back...but have government email and laptop now. Access coming soon. Have received orientation docs and WHMIS docs. Moving along :)

This is where my notes on Covid end. I wouldn't end up getting into the lab until nearly September. By mid-October I was on the road back to Ontario to return the car and spend some time with the family before flying back. Thankfully I met a few girls through the AAFC virtual meetings. Isabelle, Mackenzie, Jess, Sarah, and I had some good times with Fredericton and the 'Atlantic bubble' being less restrictive than the rest of Canada. We went camping in PEI. We took weekend trips to Halifax, where I was able to visit my aunt. I was in 6 different provinces in 2020. Wild, considering what was going on.

Over the winter, things changed. When I returned in May of 2021, New Brunswick was requiring 14 day quarantines, but unlike early in 2020, they were requiring people (if they didn't already live there and were able to quarantine privately) to stay in hotels for at least the first 7 days. Fortunately, UBC was able to help with these costs. But I was stuck in that room for 7 days, doing jumping jacks and pushups to keep sane and fit. Of course, I watched a lot of movies. The craziest thing was getting my first Covid test. Protocol was to sit in a chair inside the door. They would open the door, all dressed in clothes that reminded me of those scenes in E.T. They swabbed me and left. As far as I know, I never got Covid. The hotel got a few cases of Covid inside, shortly after I left quarantine. Then the whole program was shut down. Had I arrived two weeks later, I wouldn't have had to quarantine. It was an utter gong show. I played a lot of phone tag those two years trying to figure out how to get negative tests to the right people. The check-ins I was supposed to have in 2021 never happened. Nobody ever showed up to the house to check on me after arrival. I could have been anywhere. Both provincial and federal governments were unbelievably unprepared for this to happen. All in all, I was very fortunate to be able to travel so much and continue with my program.